

Earth's Future



10.1029/2023EF003553

Key Points:

- Time-dependent approaches are key to capture changing hazards, with diverse altered frequencies, intensities, timing or spatial occurrences
- Advances in data-driven methods and increasing availability of geodata foster new ways to detect changes in natural hazards
- Over-simplification or averaging over large scales in time or space cause severe information loss and may hide the mechanisms for change

Correspondence to:

K. Vogel, kristin.vogel@bam.de

Citation:

Vogel, K., Sieg, T., Veh, G., Fiedler, B., Moran, T., Peter, M., et al. (2024). Natural hazards in a changing world: Methods for analyzing trends and non-linear changes. *Earth's Future*, *12*, e2023EF003553. https://doi.org/10.1029/2023EF003553

Received 3 FEB 2023 Accepted 1 MAY 2024

Author Contributions:

Conceptualization: K. Vogel, T. Sieg, A. Bronstert Data curation: G. Veh, B. Fiedler, T Moran M Peter E Rottler Formal analysis: G. Veh, B. Fiedler, T. Moran, M. Peter, E. Rottler Funding acquisition: A. Bronstert Investigation: K. Vogel, T. Sieg, G. Veh, B. Fiedler, T. Moran, M. Peter, E. Rottler Methodology: G. Veh, B. Fiedler, T. Moran, M. Peter, E. Rottler Project administration: A. Bronstert Software: G. Veh, B. Fiedler, T. Moran, M Peter E Rottler Validation: G. Veh, B. Fiedler, T. Moran, M. Peter, E. Rottler Visualization: G. Veh, B. Fiedler, T. Moran, M. Peter, E. Rottler Writing - original draft: K. Vogel, T. Sieg, G. Veh, B. Fiedler, T. Moran, M. Peter, E. Rottler, A. Bronstert

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Natural Hazards in a Changing World: Methods for Analyzing Trends and Non-Linear Changes

K. Vogel^{1,2} ^(D), T. Sieg¹ ^(D), G. Veh¹ ^(D), B. Fiedler³, T. Moran⁴, M. Peter⁴, E. Rottler^{1,5} ^(D), and A. Bronstert¹ ^(D)

¹Institute of Environmental Science and Geography, University of Potsdam, Potsdam, Germany, ²Federal Institute for Materials Research and Testing (BAM), Berlin, Germany, ³Institute of Mathematics, University of Potsdam, Potsdam, Germany, ⁴Institute of Meteorology, Freie Universität Berlin, Berlin, Germany, ⁵Department of Geography, University of Innsbruck, Innsbruck, Austria

Abstract Estimating the frequency and magnitude of natural hazards largely hinges on stationary models, which do not account for changes in the climatological, hydrological, and geophysical baseline conditions. Using five diverse case studies encompassing various natural hazard types, we present advanced statistical and machine learning methods to analyze and model transient states from long-term inventory data. A novel storminess metric reveals increasing European winter windstorm severity from 1950 to 2010. Non-stationary extreme value models quantify trends, seasonal shifts, and regional differences in extreme precipitation for Germany between 1941 and 2021. Utilizing quantile sampling and empirical mode decomposition on 148 years of daily weather and discharge data in the European Alps, we assess the impacts of changing snow cover, precipitation, and anthropogenic river network modifications on river runoff. Moreover, a probabilistic framework estimates return periods of glacier lake outburst floods in the Himalayas, demonstrating large differences in 100-year flood levels. Utilizing a Bayesian change point algorithm, we track the onset of increased seismicity in the southern central United States and find correlation with wastewater injections into deep wells. In conclusion, data science reveals transient states for very different natural hazard types, characterized by diverse forms of change, ranging from gradual trends to sudden change points and from altered seasonality to overall intensity variations. In synergy with the physical understanding of Earth science, we gain important new insights into the dynamics of the studied hazards and their possible mechanisms.

Plain Language Summary According to global databases on natural hazard events and associated risks, there has been a noteworthy escalation in the extent of economic losses during past decades. It is important but difficult to distinguish and disentangle trends due to changing hazard occurrence or damage potential. Accurately quantifying altered hazards requires high-quality data sets and robust statistical methodologies. Here, we present recent progress in earth and data science toward a quantitative assessment of natural hazards in a changing world. We show that winter storms have become more frequent and more severe in Europe; that extreme precipitation in Germany shows seasonal shifts and changing intensities with regional variation; that river runoff in Central Europe is changing due to modifications of the river network, declining snowpacks, and changes in precipitation; that frequency of glacier lake outburst floods in the Himalayas have remained unchanged over the past 30 years despite rapid glacier melt and lake growth; and that earthquake activity in Oklahoma (USA) has increased with the onset of wastewater injection wells. We infer that recent advances in data science can efficiently provide new knowledge from big data sets, but interpreting these results needs a solid understanding and rather detailed analysis of the underlying processes.

1. Introduction

1.1. Natural Hazards in a Changing World

Global data bases on natural hazards show an increasing trend in the number of reported events, the number of affected people, and the amount of economic losses in past decades (CRED & UNISDR, 2020). For instance, the direct economic losses as a consequence of natural hazard events have almost doubled in the past four decades, from 1.63 trillion USD in the period 1980–1999 to 2.96 trillion USD in 2000–2019. The bulk of the increasing damages is due to a growing exposure of people and economic assets in disaster-prone areas (EEA, 2017; IPCC, 2014). Exposure has increased the most in urban areas that have been gradually growing in size, population, and assets in the private and public sector (Hoeppe, 2016). In light of increasing losses, societies have





invested in mitigation measures to reduce their vulnerability to impacts from natural hazards. For example, Thieken et al. (2016, 2022) show that adapted spatial planning, comprehensive property-level mitigation, effective flood warnings, and a targeted maintenance of flood defense systems are effective measures to reduce losses from floods over time. Further, due to improved early warning systems and evacuation plans, the number of fatalities from natural hazard events has declined and societies have become less vulnerable in recent decades (CRED & UNISDR, 2018; EEA, 2017; UNISDR, 2011). Nevertheless, events with devastating damage and high numbers of victims may still occur, even in high-income countries. For example, the disastrous flood in western Germany in July 2021 (Dietze et al., 2022; Kron et al., 2022; Vorogushyn et al., 2022) caused damages of 33 billion EUR and killed 189 people (Kron et al., 2022). Despite growing data bases, it remains debated how much human perturbation of natural systems, and atmospheric warming in particular, has contributed to the increase in weather-related events and associated losses (Bronstert et al., 2020; F. E. Otto et al., 2020). The Intergovernmental Panel on Climate Change (IPCC) reports changes in the rates, intensities, and spatial extent of natural hazards with growing evidence (IPCC, 2012; IPCC, 2014, 2019a, 2019b).

The above described dynamics in the natural risk components point out a shortcoming in the widely-used risk equation $R = H \cdot E \cdot V$ (B. Merz et al., 2010; Plate, 2002; UNISDR, 2011), which describes risk *R* as a product of hazard *H*, exposure *E*, and vulnerability *V* (Peduzzi et al., 2001). A core assumption in this equation is that all terms are stationary over time. Yet, we know that at least the exposure (Gueneralp et al., 2015; Hallegatte et al., 2013; Jongman et al., 2012; Rohat, Flacke, Dosio, Dao, & van Maarseveen, 2019) and vulnerability to several hazards (Byers et al., 2018; Jongman et al., 2015; I. M. Otto et al., 2017; Rohat, Flacke, Dosio, Pedde, et al., 2019; Thieken et al., 2022) have changed in past decades, as did likely the hazard itself (Blöschl et al., 2019; IPCC, 2012; Kemter et al., 2020; Kundzewicz et al., 2014; Mueller & Pfister, 2011). Consequently, a more adequate form of the risk equation needs to account for transient states, where process variables are subject to change and require a time dependent evaluation of natural hazards *H*(*t*) and societal factors *E*(*t*) and *V*(*t*):

$$R(t) = H(t) \cdot E(t) \cdot V(t).$$

However, identifying and describing transient hazards is challenging. At this point, it is important to note the difference between the triggering natural hazard event and the damage it causes. The event that is directly related to the damage caused is often colloquially referred to as a natural disaster. The focus of this manuscript is on the changing dynamics of the occurrence of natural hazard events, that is, not on the study of the associated losses. We address the limitations encountered in natural hazard analysis and propose strategies to overcome them. In addition to presenting novel data evaluation approaches, we emphasize the importance of integrating domain knowledge to ensure model credibility. Given the inherent complexity of natural hazard events, advanced evaluation techniques and a thorough understanding of the hazard under study are essential to distinguish transient states from natural variability across temporal or spatial scales. In addition, an understanding of the hazard is essential to identify environmental drivers and their potential interactions. Overlapping or counteracting processes, coupled with limitations in data quality, can mask or distort change signals. These issues are discussed in more detail in the following section.

1.2. Challenges in Assessing Natural Hazard Changes

1.2.1. Natural Variability on a Temporal Scale

Natural hazards are characterized by temporal variability and random processes that control the occurrence of individual events. For example, the occurrence of earthquakes is assumed to follow a stochastic Poisson process (van Stiphout et al., 2012), and climatological data can have an inter-annual to decadal periodicity (Frei et al., 2000; Scherrer et al., 2016). Further, temperature or runoff data include characteristics of long memory processes (Franzke, 2012; Rust, 2007; Rypdal et al., 2013), which means that values measured today show a long-range dependence on values measured in the past (Hurst, 1951). This self-similarity leads to trends in time series of climatic variables, which are difficult to distinguish from climate change signals. The differentiation of non-stationary behavior from stochastic processes requires a careful analysis of reliable long-term data.

1.2.2. Natural Variability on a Spatial Scale

Spatial variability can be observed on different spatial scales not only in the hazard potential, but also in the trends of natural hazards. Global studies reveal geographical patterns in the development of climate extremes, such as in temperature, storms, precipitation, or droughts (IPCC, 2014; EEA, 2017; Stott et al., 2016; J. Vogel et al., 2021). Floods along major European rivers both have increased and decreased in magnitude because of atmospheric warming and subsequent altered hydrological conditions (Blöschl et al., 2019; Kemter et al., 2020). Geographical variations limit the spatial transferability of time-dependent hazard models and usually require the underlying data to be restricted to a certain area.

1.2.3. Superimposing and Counteracting Factors

Several interacting factors can intensify the occurrence of potentially adverse natural processes. For instance, landslide activity might be affected by various superimposing impacts of climate change (Öztürk, 2018), such as altered frequencies of landslide-triggering precipitation (Gariano & Guzzetti, 2016), glacier retreat, and permafrost degradation (Knight & Harrison, 2009), as well as shorter snow cover duration (Beniston et al., 2003). Not only the occurrence, but also the impacts of natural hazards can be either eliminated or enhanced by the interplay of influencing factors. As an example, for flash-floods Bronstert et al. (2018) describe the interacting and amplifying effects of extreme rainfall intensity, runoff formation, soil and river bank erosion, and subsequent urban-area inundation and extreme damage. By analyzing large-scale drought hazard, J. Vogel et al. (2021) detect an increase in the joint occurrence of droughts and heat waves in the Mediterranean, which exacerbates the negative consequences of the individual events. The joint or sequential occurrence of hazardous processes complicates the quantification of changing impacts (Zscheischler et al., 2018).

1.2.4. Limitations in Data Quality and Availability

In many cases, observation periods fall short of robustly tracking changes in the frequency or intensity of natural hazards. Monitoring processes may be interrupted in regions with disputed borders, for example, in High Mountain Asia, where data are confidential and physical access to those regions can be limited for nonresidents. Those regions often provide inconsistently filled and poorly annotated databases, and international recognition of natural hazards remains scant (Leinss et al., 2020). The Himalayas, for example, still have no permanently operating weather station above 5,000 m a.s.l., where climate-driven hazards such as flash floods, landslides, debris flows, or glacier detachments can develop (Salerno et al., 2015). In other cases, extreme events can disrupt continuous measurements. Peak discharges, for instance, can damage stream gauges or exceed their recording capacity (Korup & Tweed, 2007). Historic inventories are usually characterized by reporting gaps and biases, since only the largest events or those with disastrous consequences are documented (Wirtz et al., 2014). Reinsurance companies report a more consistent record after 1980 in their catalogs of disastrous natural events, though such data bases are likely biased towards events with insured losses (Wirtz et al., 2014). Public interests and political developments may also impact the documentation of natural hazards. Climate-related data are studied more intensively nowadays, for example, in the framework of political initiatives, such as the United Nations Sendai Framework for Disaster Risk Reduction 2015-2030 (UNISDR, 2015) and the EM-DAT International Disaster Database (CRED & UNISDR, 2020). Rising awareness and more intense reporting may increase the reported number and related losses of natural hazard events (CRED & UNISDR, 2018; Gall et al., 2009). Further, technical developments improve the conditions for data acquisition, storage, and exchange. For example, modern instrumentation and changing station density impacts on the continuous measurement of atmospheric and hydrological variables, such as air temperatures, precipitation or river discharge. It is noteworthy that in some regions an increasing density is reported, while in others the density decreases, see, for example, (Fekete et al., 2012). Yet, changes in instrumentation may result in inconsistent recordings with changes in precision, resolution, and measurement position. For instance, changes in data collection of sea surface temperatures led to erroneous warming patterns in the early twentieth century (Chan et al., 2019). Due to these obstacles in natural hazard recordings, their analysis requires not only good knowledge of statistics, but also of data acquisition, data processing, and the studied process to avoid pitfalls in the data evaluation.

1.3. Data Science for Time-Dependent Hazard Assessments

1.3.1. Data Science for Natural Hazard Assessments

Despite these limitations, past years have seen a major advance in homogenizing global earth science and climate data sets. For example, global collections of daily air temperature date back to the 1850s (Brugnara et al., 2019) and enable more precise analyses of global warming. Reconstructing past events and homogenizing event catalogs have helped to establish continental-scale databases, for example, for river floods or tropical cyclones (Blöschl et al., 2017; Gudmundsson et al., 2021; Kossin et al., 2020). Such catalogs have allowed researchers to trace changing exceedance probabilities, or return levels, over many decades. In addition, remote sensing can help to independently estimate the timing, spatial extent, and some of the impacts of geomorphological and hydrological natural hazards, at least for the past few decades. As the ability to obtain, store, and analyze data has increased rapidly in the 21st century (Reichstein et al., 2019), more and more natural hazard assessments can be based on data-intensive algorithms. More satellite missions, seismic stations, and automatic river gauges, for example, have been producing data in amounts that exceed manual or visual interpretation (Gorelick et al., 2017; Marone, 2018).

Data-driven algorithms can help to recognize patterns and identify trends of natural processes that would otherwise have eluded human notice. For instance, the increasing number of seismic networks in combination with machine learning classifiers enables monitoring systems to go beyond solely detecting Earthquake tremors. Hammer et al. (2017) apply hidden Markov models, a probability-based sequential data analysis, to seismic data for an automatic detection of wet-snow avalanches in the Swiss Alps and provide information about avalanche activity which is independent from visual observations. Hibert et al. (2019) apply an automated classifier based on Random Forests to continuous seismic recordings and detect >5,000 landslides over a 22-year period in Alaska that had eluded scientific documentation. Cross-recurrence plots applied to runoff data helped to determine the (dis)similarity of rainfall-runoff events, and to quantify changes in the hydrological regime along the Elbe River, Germany, between 1901 and 2010 (Wendi et al., 2019). Marwan (2019) tied recurrence plot techniques to other environmental data, for example, to study transitions in past climates or the influence of external factors on ecological or climatic systems. The field of data science has brought forth machine learning classifiers such as random forests, support vector machines or neural networks that efficiently mine for structures in large data sets at high accuracy.

Another lively branch in data science uses Bayesian methods that solve, and predict from, multi-dimensional models under full probabilistic reasoning (Korup, 2021; Viglione et al., 2013). For example, K. Vogel et al. (2018) apply probabilistic graphical models, such as Bayesian networks, to detailed flood damage survey data to identify damage-driving or -reducing factors in the hazard component, the exposure, and vulnerability. A Bayesian treatment could enrich the current practice of natural hazard appraisals, because it informs practitioners clearly about uncertainties tied to expert judgment or data quality.

1.3.2. Potentials for Time-Dependent Hazard Models

While data science techniques continue to improve the quality of natural hazard research, the development and use of time-dependent hazard models remains relatively rare. In this study, we explore the potential of advanced statistical and machine learning techniques to identify and quantify changing hazards. To this end, we present five case studies that use rich databases and modern evaluation methods to analyze and model the evolving intensities and frequencies of different types of natural hazard events. Our overall goal is to establish the ability to characterize changes in natural hazards and project their shifting intensities under a warming atmosphere. Achieving this goal requires careful consideration of the inherent challenges, including seasonal and regional variability, interacting processes, and limited data, as outlined in Section 1.2. Each case study in this paper covers one or more of these issues and provides tools to address them. The case studies are drawn from recent publications authored by our team and were selected to show previously unknown aspects of changing hazard across a wide spectrum of hazard types. For ease of reference, Table 1 provides a brief summary of the hazard types, study regions, expected changes, methodologies used, challenges encountered, and relevant publications.

For the study at hand the examples have been tailored to adhere to the following guidelines: specify the hazard type under investigation and its anticipated changes, describe the available data and the applied evaluation technique, characterize any detected changes in the hazard, and discuss the findings within the broader context of

applicable Creativ

AGU
ADVANCING EARTH AND SPACE SCIENCES

Earth's Future

Table 1 Descrip	tion of the Case Studies Considere	d in the Individual Secti	ons of This Paper			
Sec.	Type of hazard	Study region	Expected change	Method	Challenge	References
7	Winter storms (meteorological)	Europe	Long-term trend	Temporal trend analysis	Decide on a meaningful feature selection to increase the information gain from limited data (Section 1.2.4) and improve the distinction of long-term trends from random fluctuations (Section 1.2.1)	Moran (2021)
3	Extreme precipitation (hydro- meteorological)	Germany	Long-term trend and changes in seasonality	Time-dependent extreme value distribution	Capture regional variations (Section 1.2.2) and seasonal and long-term variations (Section 1.2.1)	Fischer et al. (2018, 2019); Peter et al. (2024) ^a
4	Riverine floods (hydrological)	European Alps and Rhine region	Long-term trend and changes in seasonality	Empirical mode decomposition	Distinguish random fluctuation (Section 1.2.1) from structural change with season-dependent impacts and understand impacts of interacting components (Section 1.2.3)	Rottler, Francke et al. (2020)
۰.	Glacier lake outburst floods (GLOFs; hydrological, gravitational)	Himalayas	Long-term trend	Automated event detection and trend analysis	Improve the data basis (Section 1.2.4) for the statistical evaluation of long-term trends (Section 1.2.1). Quantify regional variations (Section 1.2.2). Understand impacts of melting glaciers (Section 1.2.3)	Veh et al. (2018); Veh, Korup, von Specht, et al. (2019); Veh et al. (2020)
9	(Induced) earthquakes (geophysical)	Oklahoma (US)	Near-term transition	Automated change point detection	Detect significant changes in a stochastic process (Section 1.2.1)	Fiedler, Hainzl et al. (2018); Fiedler, Zöller et al. (2018)
aPlease	note that M. Fischer changed her n	name to M. Peter.				

23284277, 2024, 5, Downloaded

from https://agupu

.wiley

the underlying physics. The method descriptions are restricted to the essential aspects, aiming to provide a fundamental understanding of the utilized concepts. The methodological details, for which we refer to the original studies, are not the primary focus of this paper. Instead, we approach the case studies from a new perspective to support two hypothesis:

- 1. Natural hazards can be subject to dynamic changes that require a time-dependent assessment of their risk potential. These changes take many forms, including gradual or abrupt shifts in frequency, intensity or seasonality.
- 2. Advanced statistical and machine learning techniques, coupled with increased data availability, are improving time-dependent hazard assessments. However, the diverse nature of evolving hazards and their specific requirements underscore the need for close collaboration between earth sciences and data sciences to develop innovative analytical methods for reliable and comprehensive knowledge acquisition.

The presented case studies demonstrate a dynamic feedback between data science and physically oriented research branches, involving disciplines such as statistics, machine learning, seismology, hydro-meteorology and geomorphology. This interdisciplinary approach significantly enhances the selection of appropriate evaluation techniques and ensures a robust and high quality study design.

2. Trends in European Winter Windstorms

2.1. Motivation

The European Environment Agency reports an averaged annual economic loss of 4 billion EUR caused by meteorological events between 1980 and 2020 in EU member states (EEA, 2022), based on the CATDAT data set. In 1999 the winter storm Lothar alone inflicted insured losses of approximately 6 billion EUR and caused 110 fatalities (MunichRE, 2015). European winter storms form over the North Atlantic and attain maximum wind speeds of 140-200 km/hr and even up to 250 km/hr in exposed coastal areas and higher mountains. A sound understanding of the factors that affect their genesis, trajectory, and evolution can aid in efforts to predict them. Thus far, M. Donat et al. (2011) have shown a statistically significant positive trend in the frequency of gale days and extreme wind speed percentiles, but this signal is only present in select parts of Northern Europe. Additionally, Befort et al. (2016) have found positive trends in European windstorm counts, but not in extreme cyclones. With regard to storm magnitudes, Ulbrich et al. (2009) show that parts of Europe can expect to see an increase in the frequency of extreme storms in warmer climates in the future. Windstorm magnitudes in particular were first introduced by Leckebusch et al. (2008) and have been used for a number of specific European case studies such as Pirret et al. (2017), but to date windstorm severities have not been studied systematically across a large set of storms. Here, we study windstorms which are tracked as persistent synoptic features (Leckebusch et al., 2008). Although windstorms by definition are "extreme" systems, their severity varies significantly between storms (Moran, 2021). Thus, to gain a comprehensive grasp of evolving windstorm effects, considering only their frequency isn't enough. It's crucial to also integrate their strengths, particularly concerning potential losses. Therefore, our goal is to analyze persistent trends in European winter windstorm severity from 1950 to 2010, using ERA-20C reanalysis data.

2.2. Data and Method

We use surface wind speeds from the ERA-20C reanalysis data set (Poli et al., 2016) in the period 1950–2010. This period is chosen as it is long enough to capture the decadal variability of windstorms, and the methods used to track them generalize well between data sets, for example, NOAA-20CR from 1950 onwards (Befort et al., 2016). The procedure used to build and study windstorm tracks is described in detail in our previous publication by Moran (2021). All variables are analyzed in the extended winter season of October to March (ONDJFM), and we only consider windstorms that passed Europe in a region between 35° S–65° N and 10° W–30° E for at least one day. Windstorms are tracked using the method developed by Leckebusch et al. (2008), which identifies windstorm properties, such as their average surface wind speed magnitudes, the location of the windstorm center, or the area of the windstorm. Together, these properties allow us to compute the dimensionless Storm Severity Index (SSI) of a storm over its lifetime, defined as



Figure 1. Storminess (black) and windstorm counts (blue) in the period 1950–2010, extracted from ERA-20C data. The dots correspond to the value of the extended ONDJFM winter in year/year + 1. Black labels are notable hazardous windstorms in the respective ONDJFM winter. Both trends (dashed lines) are significant to p < 0.05 using a *t*-test.

$$SSI = \sum_{l \in T} \sum_{k \in K} \left[\max(0, v_{k,l} / v_{k,98} - 1) \right]^3 \times \frac{A_k}{A_0},$$

where k is a rectangular area of interest, t is the 6-hourly time step, and A_k is the area (in km²) of the data's grid box in the T159 spherical-harmonic grid. A_k is normalized by a reference area A_0 , which corresponds to an area of 150,000 km². We only aggregate SSI throughout the windstorm's lifetime T and, at each time step, the total area covered K. The SSI accounts for scaled surface wind speeds $v_{k,t}$ that exceed the local 98th percentile $v_{k,98}$, where the percentile follows the assumption that storm damages occur on 2% of all ONDJFM days of the time period (Palutikof & Skellern, 1991). From this, we can construct the annual dimensionless storminess metric by summing up the SSIs of each windstorm in each extended winter season. We choose storminess over windstorm counts because storminess incorporates the severity of the storms in a given year. As storm severity is expected to change with a warming climate (Ulbrich et al., 2009), we compare linear trends and R^2 values between the counts of the windstorms and the total integrated storminess.

2.3. Results

In Figure 1, storminess shows a stronger trend compared to windstorm counts, corroborated by their R^2 values of 0.13 and 0.05, respectively. Their slopes correspond to 10-year changes of 9.10 storminess units and 0.94 windstorms, respectively. Moreover, years with peaks in storminess are those where windstorms with high damages occurred, such as Daria (1989/1990), Anatol (1999/2000), Lothar (1999/2000), Martin (1999/2000), or Kyrill (2006/2007); each of these storms were responsible for insured losses between 2 and 6 billion EUR and between 20 and 110 fatalities (MunichRE, 2015).

Storminess is correlated with windstorm counts, with a value of 0.71 (significant to p < 0.05 using a *t*-test), yet, years with peaks in storminess do not necessarily correspond with peaks in windstorm counts, which we observe, for example, in 1989, 2004, 2007, and 2008. Thus, storminess provides a different picture of the potential impacts of windstorms in a given season. While a year might appear to have lower than average numbers of storms, it

could have a high storminess, as a few individual significant windstorms can have a much greater impact. This demonstrates that storminess is an advantageous metric to use over the conventional windstorm counts. For a comprehensive results analysis, please see Moran (2021).

2.4. Summary and Conclusion

This study analyzed extreme windstorm events over Europe. Such events are of high relevance in particular for built-up areas and forested areas, and have created an enormous economic loss as well as a considerable number of fatalities across Europe. In the warming climate, it is of high importance whether the number and severity of such events might increase. In this regard, we analyzed the development of the frequency and strength of such events. We applied a linear trend analysis, which is the least complex form that can capture gradual monotonic change. The underlying data set encompasses available meteorological re-analysis data for Europe and the Northern Atlantic for the period 1950–2010. We constrained those data to the winter period (October-March), because this is the windstorm season over Europe. In general, we found that the winter storms increased over Europe. This is particularly visible when analyzing - rather than just the storm counts - the storminess index, which includes the severity of the storms in a given year and integrates the counts with their surface wind speed anomalies. The storminess shows a stronger trend compared to the windstorm count. Thus, the attribution of potential windstorm hazards is clearer when using the storminess metric instead of windstorm frequency. This can be used to give stakeholders in reinsurance or public policy more accurate information in terms of how impactful windstorms will be today and in the near future. This method can also be adapted to seasonal or decadal timescales, as well as to other regions. The method is also applicable to wind information gained from climate model simulation and therefore applicable for information regarding wind conditions in a future climate.

3. Trends and Seasonality of Daily Extreme Precipitation

3.1. Motivation

Extreme precipitation can have adverse impacts on societies, such as harvest losses or increased accident risk in road traffic. Furthermore, heavy rainfall can trigger cascading hazards, such as riverine floods, flash floods or landslides. With ongoing atmospheric warming, extreme precipitation intensity events could become more frequent and more intense (Bürger et al., 2019; Bürger et al., 2021; IPCC, 2012; M. G. Donat et al., 2016), given that a warmer atmosphere has a higher storage capacity for water vapor (Vergara-Temprado et al., 2021). Despite decades of rainfall measurements, notions of historic trends in extreme rainfall remain vague, possibly due to a lack of methods to quantify such changes. The IPCC AR5 (IPCC, 2013), for example, concludes that "are precipitation (such as the highest annual daily precipitation total) events were likely to have increased over regions with sufficient data since the late nineteenth century" (p.213).

Besides inter-annual changes in intensities and frequencies, extreme precipitation might also change seasonally (Konapala et al., 2020). Alexander et al. (2006) found that indices of extreme daily precipitation suggested a global increase in extreme precipitation from September to May. Yet, there was no trend detectable during summer (June-August). These results are consistent with the findings from Zolina et al. (2008), who analyzed seasonal trends in the 95th and 99th percentiles of daily precipitation totals in Germany. They found positive linear tendencies for winter, spring, and autumn, though summer had mostly negative trends. However, the trends were in part neither significant nor coherent in space. Looking at sub-daily time-steps, that is, hours or even shorter, analyses in Germany, Austria, and Switzerland have revealed that high-intensity but short rain storms have become more frequent in the past 35 years according to rain station data (Bürger et al., 2014, 2019; Mueller & Pfister, 2011). The trends in intensity are increasing the most between July and September. The frequency of such high-intensity rainstorms may have doubled or even tripled over a period of a few decades. A change in the seasonal cycle of extreme rainfall might change flood hazard, such that periods of extreme precipitation overlap with high water levels in rivers from snow melt (Berghuijs et al., 2019; Vormoor et al., 2015). Furthermore, a shift in seasonality could also affect crop yields, as plants are sensitive to changing precipitation during early growing stages (Derbile & Kasei, 2012; Rosenzweig et al., 2002). Understanding changes in the seasonal cycle of extreme precipitation is thus vital for a better adaptation to the impact of climate change. Here, we show an approach to quantify temporal and seasonal changes in extreme precipitation.



3.2. Data and Methods

We adopt the block maxima approach from extreme value theory (Coles, 2001) to describe the magnitude and occurrence probability of extreme precipitation. This method extracts the maximum precipitation within a given block. For this analysis, we use a block size of 1 month. The probability distribution of the block maxima can be described by the generalized extreme value distribution (GEV)

$$G(z \mid \mu, \sigma, \xi) = \exp\left\{-\left[1 + \xi\left(\frac{z - \mu}{\sigma}\right)\right]^{-1/\xi}\right\}$$

with {z: 1 + $\xi(z - \mu)/\sigma > 0$ } and the location parameter μ , the scale parameter σ and the shape parameter ξ . Instead of fitting a GEV with independent parameters to each month, we model the seasonality using harmonic functions in the form of linear models within the parameters of the GEV (Maraun et al., 2009; Mínguez et al., 2010; Rust et al., 2009). This approach offers more reliable monthly return levels and reduces model complexity. Simulations showed that including the seasonal cycles also offers more robust estimates of annual return levels (Fischer et al., 2018). To address inter-annual changes of seasonality, we extend the approach by a seasonal-temporal component (Peter et al., 2024). Conceptually, the time-dependency of the three GEV parameters in our model is represented by the following equation:

$$\theta_{\text{vear, month}} = \theta_0 + f_1(\text{month}) + f_2(\text{year}) + f_3(\text{month, year}),$$

where $\theta_{\text{year, month}}$ denotes the location, scale, or shape parameter for a specific year and month. The model includes a constant offset θ_0 , followed by three approximating functions, f_1, f_2 , and f_3 , each containing a different set of predictors. To capture the seasonal cycle, we utilize a series of harmonic functions for f_1 . For example, the first harmonic order can be represented as $f_1(\text{month}) = a \cdot \sin(\omega c_i) + b \cdot \cos(\omega c_i)$, where $\omega = 2\pi/365.25$ is the angular frequency, c_i the center day of month t = 1, ..., 12, and a and b are coefficients to be estimated. The terms $\sin(\omega c_i)$ and $\cos(\omega c_i)$ serve as the predictors. To account for the changes in extreme precipitation over time, we employ Legendre polynomials up to order five on the years for f_2 . The interaction term f_3 captures the change of the seasonal cycle with the years and is a product of the predictors for intra- and inter-annual variations. This term helps to account for any joint effects between the monthly and yearly variations. We refer readers to Peter et al. (2024) for detailed expressions and descriptions of the seasonal-temporal approach and restrict the paper at hand on presenting the overarching concept. Analogous to the seasonal-temporal model, we present a seasonal-spatial model in Fischer et al. (2019) to incorporating spatial variability of extreme precipitation. The seasonal-spatial approach enables return level estimation at ungauged sites and can further enhance accuracy.

We apply the seasonal-temporal model to monthly maxima from 519 long time series of daily precipitation sums in Germany obtained from the German Weather Service (Deutscher Wetterdienst, DWD (2022)). The stations are roughly equally distributed in space across the country and each of it covers at least the period 1941–2021. To select appropriate seasonal-interannual models for each station, we employ a step-wise forward regression approach using the Bayesian Information Criterion (BIC). This method allows us to identify the necessary predictors that best explain the variations in the data. After determining the final model setup, we estimate the coefficients for the model. Once the model is established, we use it to calculate return levels (quantiles of the GEV distribution for a given non-exceedance probability p) for different return periods. The return levels represent the expected precipitation magnitude that is exceeded, on average, once within a specific return period T, where T = 1/(1 - p).

3.3. Results

Implementing the aforementioned method uncovers that 65% (n = 338) of all evaluated stations encountered a historic change in daily extreme precipitation, as we comprehensively elaborate in Peter et al. (2024). Most of these stations show a change in the seasonality (f_3), in the form of an altered amplitude or a shifted phase, rather than a long-term change in heavy rain over time (f_2) with a consistent change for all months. Analyzed linear trends are in general weak, only 15%–35% (depending on return period and month) show changes of more than 5% from 1941 to 2021. A rise in return levels dominates especially in June. The changed seasonality is mainly characterized by an increased amplitude, later extreme precipitation events for shorter return periods (<10 years),





Figure 2. Monthly maxima of daily precipitation sums shown as Box-Whisker-Plots at three different stations in Germany and the 100-year return level of the first (dotted) and the last year (solid) of the observation period as well as for 1941 (dashed) (top row). The bottom row shows the temporal course of the monthly 100-year return levels.

and earlier extreme events for longer return periods (>10 years). Although changes are spatially very different, regions with similar and more pronounced properties stand out: one in southern Germany exemplified by the station *Rain am Lech* and one in the center of Germany represented by the station *Wesertal-Lippoldsberg*. Those observations and 100-year return levels are shown in Figure 2 (left and mid). The station *Mühlhausen/Oberpfalz-Weihersdorf*, as well located in the south of Germany, illustrates a phase shift to earlier times in the year (Figure 2, right).

We refer to Peter et al. (2024) for a more comprehensive and detailed analysis of the data evaluation. For the current paper, we focus on the stations presented in Figure 2, to illustrate different forms of changing extreme precipitation in Germany. At *Rain am Lech* the threat due to rare extreme daily precipitation has increased in all months of the year. The 100-year return level has risen about 58% from 1899 to 2021. The seasonal cycle at the station *Wesertal-Lippoldsberg* shows a changing amplitude with intensified events in summer and moderated in winter. This record benefits from the inter-annual modeling (see Peter et al. (2024) for model verification), although summerly return levels for the recent years are overestimated. Altered return levels in winter could be attributed to changes in large-scale atmospheric circulations (Willems, 2013) or changed temperature and block situations (Fauer & Rust, 2023), while a rise of return levels in summer may indicate more intense convective thunderstorms (Zeder & Fischer, 2020), accompanied by lightning, hail, and high rainfall intensity. We call for adapting hydraulic structures in all regions with increasing return levels, regardless of whether these changes are only seasonal or occur throughout the entire year. The *Mühlhausen/Oberpfalz-Weihersdorf* station shows a shift in the peak of the seasonal cycle by about 35 days from early August (1931) to the end of June (2021). Shifts in seasonality to more vulnerable times (e.g., coincidence with snow melt or growing stages of plants) can lead to potentially increased impacts on society.

3.4. Summary and Conclusion

This study analyzed extreme precipitation data regarding precipitation intensity. High-intensity rainstorms are of great relevance for a variety of hazards, such as local-scale flash floods, where they cause great damage particularly in urban areas or destroy crops and erode fertile soils in agricultural fields. We have analyzed

temporal - including seasonal - changes of extreme precipitation in Germany. We used a non-stationary generalized extreme value model to analyze the magnitude and occurrence probability of daily extreme precipitation intensity over Germany. The seasonal cycle was modeled with harmonic functions. Based on daily rainfall data obtained from the German Weather Service (DWD) from 519 rainfall stations, we processed a comprehensive data set for Germany of daily rainfall data for 80 years (1941–2021) or more. To summarize, we found in 65% of all stations a historic change in extreme precipitation. Most of these stations show a change in the seasonality and an intensification dominates especially in summer. However, trends are seasonally and spatially different, which is why a detailed spatial and seasonal analysis is recommended. For the time being, the detected trends are still weak, but the method gives consistent results and can be applied in the same manner for data, including the observations of the current years, and for higher resolved duration (e.g., hourly sums). The given approach needs an adaptation so that future return levels can be predicted in a meaningful way, since those are highly relevant for adaptation measures to climate change in urban hydrology and agriculture. Additionally, implementing other atmospheric covariates such as the global mean temperature or the North Atlantic Oscillation could offer a more physically motivated explanation of the changes in extreme precipitation.

4. Long-Term Changes in River Runoff

4.1. Motivation

Rivers spill over their banks occasionally, posing a threat to residential areas and cities. Floods account for large numbers of casualties and economic losses amounting to billions of US\$ each year (UNDRR, 2019). The 2002 Elbe flood (Kreibich et al., 2005; Ulbrich et al., 2003), the rain-on-snow flood in the Bernese Alps, Switzerland, in October 2011 (Rössler et al., 2014), and the devastating flood in western Germany and Belgium in July 2021 (Kron et al., 2022; Thieken et al., 2022; Vorogushyn et al., 2022) are only three recent examples of extremely damaging and costly floods in Central Europe. In general, extreme precipitation, excess soil moisture, and snowmelt largely control the timing, magnitude, and frequency of riverine floods (Berghuijs et al., 2019; Frei et al., 2000; R. Merz & Blöschl, 2003; Vormoor et al., 2016; Wetter et al., 2011). Changes in characteristics of river floods have been studied intensively in the past years. For example, Petrow and Merz (2009) found that flood hazard along large German rivers increased in winter, though not in summer, between 1951 and 2002. A larger sample of 4,262 hydrometric stations installed in Europe between 1960 and 2010 showed that atmospheric warming has been affecting the timing and magnitudes of river floods (Blöschl et al., 2017, 2019). Regional warming causes earlier snowmelt-induced floods across northeastern Europe in spring, and increasing autumn and winter precipitation has generated increasing flood magnitudes in northwestern Europe. Vormoor et al. (2015) project that flood seasonality in Norway could change such that rainfall replaces snowmelt as the dominant flood-generating process. In general, rising temperatures have been depleting seasonal snowpacks in recent decades (Laternser & Schneebeli, 2003; Marty, 2008; Scherrer et al., 2004). Projected warming is expected to further diminish alpine snow cover and reduce the seasonal redistribution of water from winter to summer via the build-up and melt of seasonal snowpacks (Hanzer et al., 2018; Marty et al., 2017). Furthermore, thinner snowpacks lower the potential of meltwater floods (Musselman et al., 2017). In addition, changes in the seasonality and peak values of extreme precipitation (see Section 3) could cause seasonal shifts in flood hazard.

Besides the possible impacts of climate change, the consequences or land-use change, modifications of the river network, or water resource management may significantly alter river runoff and hydro-extremes (Bronstert et al., 2007, 2023; Niehoff et al., 2002; Pfister et al., 2004). Particularly in high mountain regions, reservoirs for hydropower production affect river runoff, redistributing river runoff from summer to winter and thus decreasing runoff seasonality (Arheimer et al., 2017; Bosshard et al., 2013; Meile et al., 2011; Pérez Ciria et al., 2019; Verbunt et al., 2005). Throughout the twentieth century, and since the 1950s in particular, most rivers spawning from the Swiss and Austrian Alps have been interrupted by hydropower dams (Wildenhahn & Klaholz, 1996).

4.2. Data and Methods

The existing knowledge on the timing and magnitudes of floods in river basins largely hinges on time series of gauged discharge. River discharge data offer the longest available hydro-climatological record, for some locations dating back to the nineteenth century. Distinguishing between natural variability and long-term climate trends demands very long and high-quality recordings (Auer et al., 2005; Begert et al., 2005; Rottler et al., 2019; Scherrer et al., 2013; Vincent et al., 2002). Previous investigations have mostly resorted to linear trends for

estimating climate-driven changes in flood magnitudes. However, we call for analytical tools that can resolve non-linear responses in flood discharges to understand long-term changes in river runoff and underlying mechanisms.

We investigate discharge time series from four gauging stations located in Central Europe. The stations are part of the hydrometric observation networks managed by the water authorities in Germany and Switzerland. Their recordings are regularly checked to ensure high quality and reliability. The selected gauges offer a wealth of daily-resolution discharge data dating back to at least 1869, which can be obtained from the Global Runoff Data Centre (GRDC, 2020). The depicted gauges stand out for the exceptional length of their records and represent different types of flow regimes: nival, pluvial, and complex. Specifically, the Wasserburg gauge, situated on the Inn River (a Danube tributary) in Upper Bavaria, Germany, and three other gauges located in the Rhine River basin - Basel (Switzerland, representing the nival regime), Würzburg on the Main river (a Rhine tributary, typical of a pluvial regime), and Cologne (on the Lower Rhine, exhibiting complex regime). For the purpose of this paper, we exemplarily focus on the Basel gauge, which represents the alpine-dominated part of the Rhine catchment. This region experiences seasonal snow cover, making it particularly relevant for assessing climateinduced changes in snowmelt, which could significantly impact river runoff (Addor et al., 2014; Hanzer et al., 2018; Stewart, 2009; Viviroli et al., 2011). Comprehensive investigations of the other river gauges, along with high-quality temperature and precipitation data from nine Swiss climate stations can be found in our preceding study (Rottler, Francke, et al., 2020). The inclusion of temperature and precipitation records in our analysis not only supports the evaluation of the presented methodology but also aids in attributing detected changes in runoff to relevant climatic factors.

We provide a comprehensive description of the applied methodology in Rottler, Francke, et al. (2020), while in this paper we focus on presenting the general concept of the methods employed. Specifically, we utilize a complete ensemble empirical mode decomposition (EMD) with additive noise to our data. EMD decomposes a time series into oscillatory modes, effectively separating short-term signals from underlying (non-linear) trends (Huang et al., 1998; Luukko et al., 2016; Wu et al., 2007). This approach offers distinct advantages over commonly used linear methods, which often lack physical justification. By employing EMD, we attain a more flexible characterization of trend signals, as it does not require a predetermined basis function. In Rottler, Francke, et al. (2020) we provide a quantifiable comparison between the non-linear signals detected through EMD and the commonly used linear trend analysis. To ensure a highly resolved examination of longterm changes in daily resolution hydro-climatic data, we couple EMD to quantile sampling and moving time windows. Our investigation spans the time frame from 1869 to 2016. Extracting quantiles from discharge time series on a daily basis provides valuable insights into the seasonal distribution of river runoff (runoff seasonality). Runoff seasonality specifies the difference in mean runoff between seasons. Hence, low runoff seasonality describes a relatively even distribution of runoff throughout the year. High runoff seasonality points to significant variations, such as low runoff in winter and high runoff in summer. In order to assess the onset and evolution of changes, we apply EMD on a daily basis and use the resulting residuals to represent the underlying signal of change in the data (Luukko et al., 2016; Rottler, Francke, et al., 2020). To make results of different days comparable, each residual is centered by subtracting its mean. Consequently, negative/positive values indicate lower/higher runoff values compared to the average during the period from 1869 to 2016. We extend the approach to annual quantiles of discharge to investigate changes in low or high runoff characteristics (changes in quantiles).

4.3. Results

Results for the High Rhine demonstrate that in snow-dominated river basins, runoff is low in winter, when most of the precipitation is solid and stored in temporary snowpacks. Most runoff is generated in warm seasons, when snow melts and precipitation is liquid, even at rather high elevations (Figure 3a). In general, snow-dominated river basins show a strong runoff seasonality. In recent decades, runoff has increased during winter and spring and decreased from July to October (Figure 3b). Consequently, the runoff seasonality is decreasing. Accordingly, quantile values of low probability levels (<0.6) are increasing and quantiles of high probability levels (0.6–0.85) are decreasing. However, values from the highest quantiles (>0.85) have increased again in recent decades (Figure 3c).

One factor potentially altering the seasonality of alpine rivers is the construction and management of reservoirs for hydropower. In Switzerland, first hydropower stations were built in the nineteenth century and





Figure 3. Discharge of the High Rhine measured at the gauge in Basel, Switzerland, for the period 1869–2016: (a) seasonality of river runoff investigated by estimating discharge quantiles $[m^3 s^{-1}]$ on a daily basis, (b) onset and evolution of changes assessed by applying EMD on a daily basis and displaying the residuals $[m^3 s^{-1}]$ of the decomposition, and (c) changes in quantiles (determined on an annual level) using EMD and displaying the centered residuals $[m^3 s^{-1}]$ of the composition (adapted after Rottler, Francke, et al. (2020)).

almost all storage volume currently in use in the High Rhine has been made available until the mid-1970s. In accordance with Pérez Ciria et al. (2019); Brunner and Naveau (2023), our results indicate that the construction of reservoirs for hydropower productions need to be considered to explain decreasing runoff seasonality in the High Rhine over the time frame investigated (Figure 3). Building upon the weather data analyzed in (Rottler, Francke, et al., 2020), we further suspect that the increase in high quantiles of river discharge in the High Rhine (Figure 3c) predominantly results from higher precipitation rates, and less so from reservoir constructions or changes in snowmelt rates. Recent studies show that heavy precipitation events have become more intense and that precipitation has increased on average, particularly during winter (Frei et al., 2000; Lehmann et al., 2015; Mueller & Pfister, 2011; Rottler, Francke, et al., 2020; Schmidli & Frei, 2005). In addition, an upward shift of the snowline suggests that a larger area of the basin receives liquid instead of solid precipitation (Allamano et al., 2009a, 2009b; Barnett et al., 2005; Beniston et al., 2018; Gobiet et al., 2014; Rottler, Bronstert, et al., 2020; Zubler et al., 2014).

4.4. Summary and Conclusion

This study analyzed long-term changes in the runoff of rainfall- and snowmelt-dominated river basins. As both the seasonal timing and severity of either rainfall- or snowmelt-induced floods depends on the air temperature in the river basin, changes to floods in such hydro-climatological environments might occur as a consequence of regional warming. We used publicly available observations of river runoff, temperature and precipitation covering almost 150 years to quantify changes and assess underlying forcing mechanisms. In this chapter, we present results for gauge Basel. The full analysis and further details are available in (Rottler, Francke, et al., 2020). Our approach's novelty lies in the combination of non-linear trend detection, moving average trend statistics, and quantile sampling, enabling us to separate short time signals from underlying trends in highly resolved (daily), very long, and high-quality hydro-climatic data. A sequence of analytical steps extracts high-resolution signals of long-term changes and provides a comprehensive picture of existing trends. The consideration of non-linearity in changes over time facilitates the attribution of changes.

Drawing on the findings of Pérez Ciria et al. (2019); Brunner and Naveau (2023) we propose that the construction and operation of reservoirs for hydropower production might be a contributing factor to the altered streamflow characteristics in the High Rhine region, leading to a control effect on river runoff and a reduction in runoff seasonality. This aspect is discussed in more detail in Rottler, Francke, et al. (2020). The cited study also includes detailed investigations of precipitation patterns and indicates an increase in (intense) rainfall, which might contribute to the observed increase in high river discharge quantiles. Furthermore, we suspect that an upward movement of the snowline, driven by rising temperatures, results in larger areas of the basin to receive rainfall instead of snowfall. As a consequence, the importance of snowmelt as a flood-generating process will further decrease, while rainfall-triggered flood events could become more severe and frequent. The investigation of downstream river gauges (i.e., Cologne) suggests that alpine signals can propagate downstream and affect runoff in river segments far beyond the alpine area.



5. Hazard From Himalayan Glacier Lake Outburst Floods (GLOFs)

5.1. Motivation

The Himalayas host the largest reservoir of ice outside the polar regions. Yet, glaciers in the Himalayas have thinned, retreated, and lost mass since at least the 1970s (Bolch et al., 2019), and rates of ice loss have accelerated since the 2000s (Maurer et al., 2019). Ongoing glacier retreat has created space for new glacier lakes, and existing lakes have been growing from sustained meltwater supply. Glacier lake abundance and area in the Himalayas increased by 11% and 15%, respectively, between 1990 and 2018 (Wang et al., 2020). Lakes dammed by abandoned moraine dams account for one quarter of all (n = 8,300) Himalayan lakes (Maharjan et al., 2018), yet they had some of the highest growth rates in this period (Nie et al., 2017). Moraine dams are prone to failure because they are made of loose, poorly sorted sediment, possibly accommodating ice cores (Clague & Evans, 2000; Richardson & Reynolds, 2000). Steep slopes adjacent to moraine-dammed lakes could also trigger ice avalanches, rockfalls, and landslides, which could impact lakes and produce splash waves that overtop the lakes (Haeberli et al., 2017). When glacier lakes fail, several million cubic meters of meltwater and sediment can drain in devastating glacier lake outburst floods (GLOFs) (Osti & Egashira, 2009; Watanbe & Rothacher, 1996; Xu, 1988). Given ongoing atmospheric warming, glacier retreat, and lake growth, GLOFs are projected to become more frequent (Harrison et al., 2018). Yet, trends in the historic activity of GLOFs remain controversial (Carrivick & Tweed, 2016; Harrison et al., 2018; Nie et al., 2018; Richardson & Reynolds, 2000). Many outbursts may have gone unnoticed in Himalayan headwaters, limiting our knowledge on GLOFs to larger, destructive cases (Komori et al., 2012). Such inconsistencies in GLOF databases call for a more complete GLOF inventory to estimate changes in GLOF frequency, and hazard, eventually.

5.2. Data and Methods

We developed a processing chain in Veh et al. (2018) to detect previously unrecorded GLOFs from time series of Landsat satellite images. To systematically fill the GLOF inventory in the Himalayas, we first trained a Random Forest classifier (Breiman, 2001) that predicts land cover for >8,000 post-monsoon images (September to November) between 1988 and 2017. Random Forest is a popular ensemble learning classifier in remote sensing because this method combines many hundreds of independent decision trees by using bootstrapped samples of land cover classes (Belgiu & Drăguț, 2016). Random feature selection within the model reduces overfitting, while improving accuracy. We defined five land cover classes, manually labeled 12,500 training pixels with one of these classes, and trained the Random Forest model. We aggregated the votes of the individual trees and obtained an overall accuracy of 90% of the model. This high accuracy is suitable to predict the land cover for each pixel of each Landsat image in our time series. The Landsat mission has a short revisit time of 16 days, and thus allows us to detect lakes that shrank between two successive land cover maps. The partial or complete loss of water bodies is one of two key diagnostics for outbursts, judging from the 17 previously reported cases that we collated in this period. Based on this observation, we developed an automated change point algorithm that explores the stack of land cover maps for pixels that shifted from water to land (Veh et al., 2018). A second key diagnostic for GLOFs are sediment fans below shrunken lakes. We visually assessed all sites of predicted water loss for downstream sediment aggradation to complete our search for previously unrecorded GLOFs.

We then merged the inventory of previously reported cases with our newly detected GLOFs (see Tables S1 and S2 in Veh, Korup, von Specht, et al. (2019)). We tested whether this updated inventory shows a trend in GLOF frequency over time using a Bayesian linear regression. The noise model for our target variable *annual GLOF frequency* follows a student-*t* distribution. This distribution allows for heavier tails than the Gaussian distribution that is conventionally used as the likelihood function in linear regression models. The robustness is achieved by accounting for a low number in the degrees of freedom ν in the student-*t* distribution (here $\nu = 4$), which controls the shape, and thus the probability mass, in the tails of the distribution (Kruschke, 2015). Our estimates of trends in mean GLOF rates can thus account for potential effects of outliers in the response, that is, balancing years that had many or no GLOFs. We also binned the Himalayan arc into swaths of 50-km width to assess whether regional GLOF abundance correlates with glacier area (Arendt et al., 2017), glacier mass change (Brun et al., 2017), or changes in surface water (Pekel et al., 2016).

We chose the 100-year return level of outburst volume V_0 or peak discharge Q_p as an objective metric of GLOF hazard. Return levels can be interpreted as the exceedance probabilities in an extreme value distribution. To





Figure 4. Distribution and return periods from moraine-dammed glacier lake outburst floods (GLOFs) in the Himalayas. (a) Bubbles show the size and abundance of moraine-dammed lakes in $1^{\circ} \times 1^{\circ}$ bins. Triangles are previously known GLOFs, and red dots are GLOFs detected from Landsat images between 1988 and 2017. (b) Estimated return periods of GLOFs in all regions, and in the Eastern (Western) Himalayas, where regional GLOF hazard is highest (lowest). Blue numbers are the mean 100-year return level of peak discharge Q_p including the 95% highest density interval; brown numbers are the mean GLOF rate in this region. Brown ticks are historical GLOFs. The GLOF from lake Zhangzangbo in 1981 had a peak discharge of ~16,000 m³s⁻¹ (Xu, 1988) and a 100-year return period, accordingly. Figure adapted from Veh et al. (2020).

determine the 100-year GLOF discharge, we chose the Generalized Pareto (GP) model fitted to time series of GLOF discharges. We simulated such GLOF discharge time series by assuming randomly occurring events. The parameter of the respective Poisson distribution corresponds to the observed regional GLOF frequency λ (i.e., number of GLOFs per year). Outburst volume and peak discharge are based on empirical estimates of V_0 and Q_p for each lake in a given region (Cook & Quincey, 2015; Maharjan et al., 2018; Walder & O'Connor, 1997). The GP model then takes as input all values of simulated V_0 or Q_p above a high threshold in a given time series of regional discharges. We showed in Veh et al. (2020) that our GP models are robust for the 20% highest simulated discharges and used this threshold to eventually estimate GLOF return periods. Accordingly, the 100-year GLOF is the value of V_0 or Q_p that has a 1% chance to be reached or exceeded in the GP model, which is a widely used characteristic in river flood hydrology (Katz et al., 2002).

5.3. Results

Our automated search in Landsat images reveals 22 newly detected GLOFs, as detailed in our publication by Veh, Korup, von Specht, et al. (2019). We thus more than doubled the previously known GLOF count in the Himalayas since 1988 (Figure 4a). Yet, the total of 38 GLOFs shows no change in the annual frequency (Veh, Korup, von Specht, et al., 2019): the trend in our study period is $-0.0006^{+0.0415}/_{-0.0394}$ GLOFs yr⁻¹ (mean and 95% highest density interval). Given increases in lake size and abundance, the activity of GLOFs per unit of glacial lake area has even decreased in the past 30 years. This finding could point at some resilience of glacier lakes to sudden

23284277, 2024, 5, Downloaded from https

wiley

com/doi/10.1029/2023EF003553 by Fak - Bam Berlin, Wiley Online Library on [28/05/2024]. See the

outbursts. For example, lakes could have decoupled from their retreating parent glaciers and thus could have become less prone to impacts from their calving tongues.

A spatial cluster of GLOFs has formed in the Central and Eastern Himalayas (Bhutan and Eastern Nepal), in contrast to the less frequently affected regions in the northern Hindu Kush-Karakoram ranges. We provide a detailed presentation of our swath analysis in Veh, Korup, von Specht, et al. (2019). It suggests that the regional GLOF clusters correlate with both the abundance of meltwater areas (Spearman's correlation coefficient $\rho = 0.87$; Figure 4a) and the change in lake areas, calculated as the difference between two epochs before and after the year 2000 ($\rho = 0.77$). Glacier cover ($\rho = -0.18$) and changes in mass balance ($\rho = 0.26$) have little association with the number of observed GLOFs. We infer that a higher abundance and variability of glacier lakes warrants a higher GLOF count, but not necessarily an increasing rate of lake outbursts.

We chose a stationary extreme-value model to estimate GLOF return periods, given that the GLOF frequency has remained constant over our study period. As explicated in Veh et al. (2020), our investigation yielded a contemporary 100-year GLOF volume V_0 of $33.5^{+3.7}/_{-3.7} \times 10^6$ m³ (mean and 95% highest density interval), along with a 100-year GLOF peak discharge Q_p of 15, $600^{+2,000}/_{-1,800}$ m³ s⁻¹ across the entire Himalayas (Figure 4b). The regional variations of GLOF frequency and lake sizes are key determinants in our hazard appraisal. The 100-year return level of Q_p in the Eastern Himalayas is at least triple that of any other Himalayan region, because the Eastern Himalayas have the largest lakes and the most historic GLOFs. Accordingly, GLOF hazard is smallest in the western Himalayas, where GLOFs are rare and lake cover is smallest (Figure 4b).

5.4. Summary and Conclusion

GLOFs are extreme floods triggered by a geophysical process, such as a landslide, earthquake or dam failure. These floods transport large amounts of water and debris, and may have severe consequences for the river reaches and population downstream. It has been argued that the occurrence of GLOFs may increase because of regional warming, due to a rising number of glacial lakes, reduced stability of slopes above such lakes and dams forming the lake, and more frequent ice avalanches discharging into such lakes (Haeberli et al., 2017). Yet, quantifying trends in GLOF occurrence has been challenging as these floods are rarely reported, while others may drain without reporting in uninhabited valleys. This study more than doubled the number of known GLOFs in the Himalayas using an automated processing chain of optical satellite images (more than 8,000 images between 1988 and 2017). Potential GLOFs were identified by applying a Random Forest classifier for the land cover prediction and a change point algorithm to detect shifts from water to land. Yet, a Bayesian robust linear regression applied to the total of 38 previously recorded and newly detected cases did not reveal a change in the annual GLOF frequency. GLOFs occurred in regional clusters, which correlate with the abundance of meltwater areas and the change in lake areas over the last decades (1988–2017). This spatial clustering leads to large regional differences in the calculated GLOF return periods. The three stages of GLOF analysis proposed here (from GLOF detection to analyzing their frequency and estimating regional GLOF return periods) present a suitable framework for modern GLOF hazard assessment. Given a rapidly growing population, infrastructure, and hydropower projects in the Himalayas, this approach complements local expert-based hazard and risk appraisals.

6. Transient States in Seismicity

6.1. Motivation

Earthquakes and co-seismic events such as tsunamis or landslides rank among the most devastating natural hazards worldwide. Earthquakes have caused a cumulative economic loss of 661 billion USD and 747,000 fatalities between 1998 and 2017 (CRED & UNISDR, 2018). Seismicity, for example, at active faults, is assumed to be resilient against anthropogenic impacts. However, earthquakes may be triggered, delayed or advanced in response to changing boundary conditions: stress transfer from adjacent regions, for example, due to plate tectonics, movement of magma, or tides (C. H. Scholz et al., 2019), can bring a fault closer to, or away from, failure. Man-made stresses from mining, geothermal exploration, or crustal load from large hydropower dams may also initiate seismicity. The same applies to the presence of fluid flows in Earth's crust. Stress from fluids include natural causes such as increased pore pressure after rainfall (Hainzl et al., 2006), as well as man-made intrusions such as wastewater injections into deep wells. The majority of this wastewater originates from conventional

on Wiley Online Library for rules

hydrocarbon deposits and is separated during oil production. Moreover new drilling techniques for oil and gas exploitation have shed light on injection-caused earthquakes. Hydraulic fracturing ("fracking") routinely produces micro-earthquakes below M = 2, though an increase in earthquakes with higher, potentially damaging magnitudes could not be proven (Ellsworth, 2013). However, wastewater injection into deep wells has produced a notable increase in harmful earthquakes. For example, the US state of Oklahoma has attained public and political interest due to a rapid increase of seismicity since the onset of high-volume wastewater injections in 2009 (Gupta & Baker, 2017; Langenbruch et al., 2018; Montoya-Noguera & Wang, 2017), including five events of $M \ge 5$ in Oklahoma (Ellsworth, 2013; Langenbruch & Zoback, 2016). To prevent further, possibly larger earthquakes, injection volumes had to be reduced by law in 2016 (Baker, 2017).

Changes in the stress regime may not only affect the number of earthquakes, but also the ratio of large-magnitude to small-magnitude earthquakes (Amitrano, 2003; Goebel et al., 2013; C. Scholz, 1968). A decreasing ratio of large to small magnitudes can indicate an increasing stress level in Earth's crust (Urbancic et al., 1992; Wyss, 1973), for example, as a consequence of human activity. A reliable detection of changed seismicity can be challenging due to small sample sizes, binned data, and natural fluctuation. Here, we present a Bayesian approach to identify change points in earthquake numbers and magnitude distributions based on seismic catalogs. The approach is applied to seismic catalogs from Oklahoma and southernmost Kansas (US) to substantiate a rapid increase in seismicity in connection with the growing number of industrial projects and associated fluid injections.

6.2. Data and Methods

We apply a change point model to examine the footprint of fluid injections in seismic catalogs from Oklahoma and southernmost Kansas (US). The studied earthquake catalog comprises events with magnitude $M \ge 2$ between 1 January 2005 and 31 December 2019, obtained from the Oklahoma Geological Survey (OGS, 2021). The distribution of earthquakes is assumed to follow a Poisson process. The count N(M) of earthquakes of magnitude M or greater is typically described by the Gutenberg-Richter relationship:

$$\log N(M) = a - bM, \quad M \ge M_c,$$

here, M_c is a threshold above which the earthquake catalog is considered complete. The parameter *a* represents the total seismicity rate, and *b* characterizes the proportion of smaller to larger earthquakes within a specific region.

We focus on the temporal change of the two key parameters *a* and *b* and compare two model setups to investigate earthquake occurrences in a given catalog: \mathcal{M}_0 assumes that *a* and *b* remain constant over time. In contrast, \mathcal{M}_1 expects one change point with credible differences in *a* or *b* before and after the change point. We test the hypothesis of no versus one change point using the Bayes factor, B_{01} . The Bayes factor relates the probability of observing the recorded data under the assumption of no change point, $p(\mathbf{d}|\mathcal{M}_0)$, to the probability of observing the data under the assumption of one change point, $p(\mathbf{d}|\mathcal{M}_1)$,

$$B_{01} = \frac{p(\mathbf{d}|\mathcal{M}_0)}{p(\mathbf{d}|\mathcal{M}_1)}.$$

Due to the fact that the Bayes factor quantifies the evidence of the supported model and strongly depends on the choice of the prior distributions, the Bayes factor threshold is different for the parameters *a* and *b*. Synthetic tests for *a* suggest a threshold for the Bayes factor at B = 0.05, whereas B = 0.5 is a reasonable choice for the *b*-value. This means, that for example, for the parameter *a*, $B_{01} < 0.05$ can be interpreted as a decisive evidence against a model without a change point. The concept can be extended to models with several change points. Here, the Bayes factor $B_{lk} = \frac{p(\mathbf{d}|\mathcal{M}_l)}{p(\mathbf{d}|\mathcal{M}_k)}$ is used, where \mathcal{M}_l and \mathcal{M}_k denote a model with *l* and *m* change points, respectively. We provide a detailed description of the extended approach in Fiedler, Hainzl, et al. (2018).

In the second step, we use the likelihood function to determine the most likely position of the change point and check its significance ($\alpha = 0.05$). This approach avoids smearing effects of the commonly applied moving window approach (Kulhanek, 2005; Nuannin et al., 2005), where the *b*-value is estimated from sliding time windows with predefined window length and step size. Subsequently, the detected change points are visually compared with the onset of fluid injection wells (OCC, 2021), and a correlation analysis is performed.

med by the applicable Creat





Figure 5. Locations and occurrence times of transients for models with one, two or three change points, compared with approval dates of injection wells between 2005 and 2019 in Oklahoma. Wells that had an approved injection volume >10,000 barrels per day are black. (a) Map of the first occurrence of estimated transitions, (b) latitude-time plot, and (c) time-longitude plot with all estimated change points and injection wells. For all transients, it is illustrated whether an increasing or a decreasing rate is given.

The refined location of changes in the magnitude-frequency distribution can be used not only to analyze humaninduced earthquakes, but also to detect natural transitions in the stress regime. For example, in Fiedler, Hainzl, et al. (2018) we study possible precursor signals of large magnitude events. Decreasing *b*-values have been observed before major earthquakes (Schorlemmer et al., 2005; Spada et al., 2013; C. H. Scholz, 2015), but the value of changing *b*-values to predict large earthquakes is debated (Imoto, 1991; Kamer & Hiemer, 2013; Nakaya, 2006; Nanjo et al., 2012; Smith, 1981).

6.3. Results

Figure 5 shows a considerable increase in high-volume injection wells in Oklahoma after 2010. This onset is accompanied by change points toward an increasing seismicity rate, starting with a few change point locations in 2010 and leading to a high concentration of change points toward increased seismicity between 2013 and 2015. We analyze the correlation between the change points and the onset of injection wells in Fiedler, Zöller, et al. (2018) and find it to be especially pronounced for high-volume injection wells. This correlation is consistent with the results of Langenbruch and Zoback (2016), as well as the findings of transients between 2008 and the end of 2015 by Gupta and Baker (2017) and Montoya-Noguera and Wang (2017). About 25% of all time series suggest two or more change points. The later change points are usually associated with a decreased seismic activity, possibly related to reduced injection volumes (Langenbruch et al., 2018). From around 2016, the majority of change points reflects a decreasing seismicity rate. This indicates a positive effect of the prohibition of large injection volumes in 2016.

6.4. Summary and Conclusion

This study analyzed earthquake occurrences in Oklahoma (US) with regard to suspected changes in seismicity as a consequence of the growing number of industrial projects. Earthquakes belong to the most damaging natural hazards worldwide. Their occurrence can be favored by man-made stresses, for example, from mining, geothermal exploration or hydraulic fracking. While most human-induced earthquakes are limited to small-magnitude events, the US state of Oklahoma reported a rapid increase in seismicity, including events of magnitude five or higher, since the onset of high-volume wastewater injections in 2009. We applied a change point detection algorithm to pinpoint transitions in the magnitude-frequency distribution of earthquakes recorded in comprehensive catalogs from Oklahoma and southernmost Kansas from 1 January 2005–31 December 2019. A Bayes factor that compares the performance of models with no, one or several change points is used to derive the number of transitions. Subsequently, the likelihood function is used to locate the most likely change point(s). A correlation analysis of the identified change points with the onset of injection wells confirmed an increase of



seismicity as a consequence of high injection volumes. We further found a cluster of change points with decreasing seismicity around 2016, which indicate a positive effect of the injection reduction mandate from 2016. In contrast to trend analyses or the commonly applied moving window approach, the change point algorithm has no smearing effects and is suitable for detecting rapid modifications of the studied system. The precise determination of changed seismicity simplifies the detection of the seismic response to naturally or anthropogenically changed stress regimes and improves the root-cause investigation.

7. Discussion

7.1. Need and Potential for Time-Dependent Hazard Models

Table 2 summarizes the five case studies presented, highlighting the specific challenges of the hazards studied and how these were addressed through tailored approaches. This process has led to new ways of evaluating data and the development of innovative model designs, providing new insights into changing hazard potentials.

In summary, the individual studies support the hypotheses set out in Section 1.3:

- 1. Identified changes in hazard potential underscore the need for time-dependent hazard models to capture dynamics (e.g., in frequency, magnitude, space-time variability) of hazard events, and, thus, contribute to the dynamics of the associated risks. Significant differences in the manifestations of change are evident, whether hazards evolve gradually or abruptly, show local increases or decreases, or undergo seasonal shifts.
- 2. Accordingly, data evaluation methods must be adapted to the specific nature of the hazard under investigation. The choice of method used should be based on expected changes and problems encountered in data analysis, rather than on personal preference or popularity of the method. The studies presented are examples of how an efficient study design was developed through close collaboration between data and earth scientists. This facilitated the development of time-dependent hazard models and provided new insights into the hazard type studied.

In the following, we summarize how the presented hazard analyses pave the way for better quantification of transient hazards by addressing the aspects outlined in Section 1.2, namely natural variability on temporal and spatial scales, overlapping and counteracting factors, and limitations in data quality and availability. These approaches serve as showcases and inspiration for how data science techniques can be used to overcome existing challenges in natural hazard assessment.

7.2. Challenges in Assessing Natural Hazard Changes

7.2.1. Natural Variability on a Temporal Scale

Understanding the intrinsic temporal variability in natural hazards is key to identifying transient states. For instance, meteorological dynamics can be characterized by seasonal variations. Annual values, such as annual mean temperature or annual mean precipitation, ignore seasonal effects and may hide, for example, an increase or decrease in seasonal amplitudes or timing. Average values (e.g., annual rainfall values) may also mask transient features in shorter intervals, such as for example, peak rainfall intensity in a sub-hourly time scale. Sections 3 and 4 use oscillating functions or modes to capture seasonal behavior. By identifying and distinguishing seasonality from long-term trends, we receive more distinct information for both components and, in addition, are able to detect shifts in seasonality. Another often assumed form of temporal variability, namely the random occurrence of independent events, is typically modeled as a stochastic Poisson process. This is illustrated in Section 6 for modeling the temporal occurrence of earthquakes. Poisson processes can also be used to estimate return levels of outburst floods, given that associated triggers such as earthquakes, landslides, or ice avalanches can be considered as randomly occurring and independent events (Section 5).

Both the frequency and the intensities of events are prone to change. Hazard assessments are typically interested in the extreme or largest events in a given period, as those are associated with the largest impacts. Extreme events, such as severe wind storms (Section 2), heavy rainfall (Section 3), very high or low river runoff values (Section 4), or large glacier lake outburst floods (Section 5) may be defined as the highest (or lowest) quantiles of reported intensities. This allows us to capture the tails of the probability distribution, which are usually of high relevance with regard to the damage potential. The occurrence of extreme events can be tied to extreme value distributions in order to model the probabilistic behavior of rare events (Section 3 and 5).

7.2.2. Natural Variability on a Spatial Scale

Spatial variability and the peculiarities of different hazard types limit the transferability of hazard models and the joint evaluation of data from different regions. Section 3 emphasizes that return levels of extreme precipitation in Germany vary on local and regional scales. While some regions show a general increase in extreme precipitation intensities, others show a forward or backward shift in seasonality, an increase in seasonal amplitudes, or even decreasing intensities. Further, Section 5 shows a regionally contrasting pattern of glacier lake abundance, which controls the likelihood of observing a lake outburst flood. Both studies suggest models that can account for regional and local differences in magnitude and frequency of the underlying process. Extending Section 3, we refer to Fischer et al. (2019) and Ulrich et al. (2020), who show how the location can be included in modeling extreme precipitation. This approach is especially beneficial for rare events and small data sets, since data from neighboring stations is used to train the model. Section 4 showed that the overall/large scale hydrological runoff dynamics is resulting from a combination of different hydrographical regions and altitudes of the Rhine River basin. For understanding the whole region's dynamics it is essential to understand the mechanisms at the sub-regional scale.

7.2.3. Superimposing and Counteracting Factors

Superimposing or opposing effects complicate the interpretation of observed results. For instance, the hazard of flooding could increase due to seasonal shifts in precipitation, if extreme precipitation becomes more likely to coincide with snow melting. Rising temperatures can have an opposite effect (Section 4). In the European Alps, rising temperatures reduce snowpacks and consequently decrease the potential for meltwater floods. Moreover, human impacts, such as changing land use or reservoir construction, can hide less pronounced signals, such as changes in snowmelt-induced river runoff. This example illustrates that mechanistic process knowledge is essential to interpret the results and infer possible consequences. A purely data-driven analysis that ignores the interplay of contributing factors may fail to draw relevant conclusions. For example, an unchanged GLOF frequency (Section 5) may point at an unchanged hazard in an unchanged environmental system. Yet, this is in contrast to the observed growth of glacier lakes in number and size. At the same time, GLOF-triggering mechanisms might be affected by changed environmental conditions, for example, the number of lakes affected by ice avalanches may decrease due to glacier retreat.

7.2.4. Limitations in Data Quality and Availability

Depending on the hazard and region studied, aspects of scarce or inconsistent data can be of high relevance. Especially remote regions, such as high mountain areas in general and the Himalayas in particular, suffer from poor instrumentation and limited data. For instance, considering the occurrence of GLOFs, data recordings are not only incomplete, but also biased towards large and destructive events. Remote sensing can help to systematically detect, and fill inventories of gravitational and hydrological hazard events without drawing on field work or eyewitness reports. In Section 5, the time series analysis of satellite imagery helped to compile a more complete database of GLOFs and to reduce the observation bias. As a result, it increased the reliability of the statistical trend analysis. In contrast, the remaining studies make use of permanent and continuous monitoring, which form the basis for robust statistics on seismic data (Section 6), meteorological data (Sections 2, 3 and 4), and discharge data (Section 4). The amount of data exceeds manual inspection and requires automated data processing with a careful selection of explanatory variables and model designs. Section 2 illustrates the benefit of including domain knowledge in the feature selection to retrieve more information when analyzing complex data. Compared to commonly used windstorm counts, the newly developed storminess metric shows a more distinct signal of the increasing hazard arising from European winter storms.

7.3. Data Science for Time-Dependent Hazard Assessments

In summary, to address aspects of natural variability, interfering impacts, and data quality, the analysis of natural hazards requires the integration of knowledge about the data acquisition and processing, a physical and mechanistic understanding of the studied process, and methodological expertise in applicable algorithms and potential pitfalls of data evaluation. As a consequence, natural hazard analyses benefit from a dynamic feedback between different branches of research, including statistics, machine learning, and physical models, which describe the seismological, hydro-meteorological, ecological or geomorphological processes. To exploit the available data



Earth's Future

	n Detected change	 Increasing trend not only in storm counts, but also in storm severity; both are captured in storminess 	 Jong-term trends No overarching trend, pronounced seasonal and spatial variations Changing seasonality with altered ampli- tudes or shifted phases is more often than uniform changes throughout the year 100-year return levels tend to increase 	n combined with • Gradually decreasing snowpacks reduce the tistics and quantile flood hazard from meltwater-induced flooding flooding tion signals from • Rainfall-triggered flood events could nprehensive hydro- become more frequent oution	Jy unknown casesNo apparent change in annual GLOF fre- nown GLOF count quency, despite growing glacier lakesnown GLOF count periods- Decrease in GLOF activity per unit of glacial lake areaur periods- Decrease in GLOF activity per unit of glacialur variations in a variations- Large regional variations	 points improves Seismicity increases with the onset of high- micity to its causes volume wastewater injection wells sments about Seismicity decreases with reduced injection volumes 	challenges. It also outlines the innovative aspect of the study design
	Innovatio	 Severity index captures v counts and surface wind 	 Captures seasonal cycles and changes in seasonali Improves the prediction of 	 Non-linear trend detectio moving average trend sta sampling Separation of high-resolu long-term changes in con meteorological data Explanatory change attrift 	 Detection of 22 previous more than doubles the kr Estimation of GLOF retu Quantification of regiona GLOF hazard 	 Precise location of chang allocation of altered seist Allows probabilistic statt change point occurrence Allows detection of mult 	address and overcome these
10 6	Tailored approach	Trend analysis with new storminess metric that captures frequency and magnitude	Non-stationary generalized extreme value models with harmonic functions to capture the seasonal cycle	Empirical mode decomposition coupled to quantile sampling and moving time windows applied to daily hydro-climatological data	Random forest classifier applied to Landsat satellite image combined with an automated change point algorithm; robust linear regression and Generalized Pareto models	Bayesian approach for model selection (number of change points); maximum likelihood approach for change point location	ented hazard studies, and the approaches taken to zard.
ented in Sections 2	Addressed major challenges	No systematic study of windstorm severity	Lack of methods to detect (seasonal) changes in extreme precipitation	Lack of an approach to extract high- resolution and non- linear signals of long-term changes	Incomplete GLOF inventory with reporting bias	No automated data processing for change point detection	allenges in the prese to the changing ha
Table 2Summary of the Case Studies Press	Case study	Section 2: European winter storms	Section 3: Extreme precipitation (Germany)	Section 4: River runoff in alpine catchments (Alps and Rhine region)	Section 5: Glacier lake outburst floods (Himalayas)	Section 6: Induced Seismicity (Oklahoma, US)	<i>Note</i> . This table summarizes the ch and the main findings with respect

most effectively, their analysis needs a careful selection of suitable evaluation techniques with consideration of the research question and the underlying physics. Especially machine learning approaches may not be completely traceable and at some points resemble a black box. Moreover, random coincidences or artifacts in the data can lead to unexpected results. To diminish misinterpretation, the study should be designed to prove or disprove a certain hypothesis, such that "high wastewater injection rates increase the seismic activity" (Section 6). Consequently, a useful study design demands physical understanding of the underlying process and knowledge about applicable algorithms.

In the presented hazard analyses, each study was designed according to the physical understanding of how frequencies, magnitudes, and return periods of the studied hazard might have changed. We applied linear models to express the belief that hazard frequencies or intensities might have changed more gradually, and less abruptly or in another more complex form. For instance, in Section 2 the linear model confirms an increasing trend in the number and intensity of winter storms in Europe. Surprisingly, in Section 5 the linear model shows no trend in GLOF frequency in the Himalayas, despite a growth of glacier lakes in number and size. This finding is difficult to explain, and its interpretation requires the consideration of trigger mechanisms and the underlying physical processes. The models in Sections 3 and 4 are not limited to linear trends, but also account for seasonal behavior and changes in seasonality. They identify local trends and seasonal shifts for extreme precipitation in Germany (Section 3) and river runoff in the European Alps (Section 4). In contrast to linear models, a change point model is preferred if the studied hazard is expected to be subject to more sudden changes. Unlike a trend analysis or a moving window approach, the change point model can capture sharp responses to abrupt system changes. In Section 6, a change point model helps to detect the onset of increased or decreased seismicity as a consequence of changed wastewater injection. In each studied research question, the custom-fit model selection is key to provide new findings from the available data.

8. Conclusion

We have presented five case studies to illustrate the multifaceted nature of changing natural hazards in recent decades. It is important to realize that such changes can become apparent in multiple data characteristics, for example, in frequency, magnitude, temporal or spatial pattern (either as an organized or random type of variability), or in a gradual/linear or sudden/abrupt manner. Our selection includes different hazard types and different manifestations of change: The increasing trend in European winter storms could be highlighted by showing not only an increase in the number of storms, but also in their severity (Section 2). Different forms of change could be observed for extreme precipitation in Germany. Increasing and decreasing trends in daily precipitation extremes could be identified, as well as shifts and changing amplitudes of the seasonal cycle. In addition, the prediction of return levels could be improved (Section 3). Decreasing snowpack, changes in precipitation and river engineering are leading to a change in the seasonality of river discharge in Alpine catchments, with a reduced risk of meltwater-induced floods but an increased frequency of rainfall-induced floods (Section 4). Despite the increasing number and area of glacial lakes in the Himalayas, no change in annual GLOF frequency has been observed over the last 30 years. However, spatial variations in GLOF hazard are significant and highly correlated with the abundance of meltwater areas and the change in lake areas over the past decades (Section 5). The identification of change points in seismicity data from Oklahoma (USA) and their correlation with the onset of wastewater injection wells improves the attribution of increased seismicity to the onset of high-volume injection wells (Section 6). The synthesis of the case studies not only demonstrates the need for time-dependent hazard models to capture changing hazard potentials, but also illustrates the different ways in which change can manifest itself, such as gradual versus abrupt changes, altered amplitudes, or seasonal shifts.

Scientific advances in data-driven methods and the increasing availability of climate-related and environmental data have facilitated in-depth analysis of large data sets beyond human interpretation. This is paving the way for new opportunities to identify and quantify changes in natural hazards. However, each hazard assessment has different needs and shortcomings, requiring tailored methodologies. The case studies presented provide strategies to address specific shortcomings, such as variability on temporal and spatial scales, interacting processes, or limited data quality. They serve as examples for similar scenarios and suggest assessment techniques tailored to the specific nature and expected change of the hazard under consideration: Linear models serve to identify gradual trends, oscillatory functions adequately describe seasonal variations, extreme value distributions are effective in modeling changes in the most hazardous events, and change point detection techniques are used to identify sudden shifts and improve attribution to driving forces of change.

23284277, 2024, 5, Downloaded from https://agupubs.onlinelibrary

As a final conclusion, we state that

- The presented combination of studies demonstrates the need for an adequate and customized analysis and model selection. The choice of an appropriate study design is critical for reliable and efficient hazard assessment under change. When applied correctly and supported by a mechanistic/physical understanding of the governing processes, this will enable to disentangle the dynamics about the hazard type under investigation.
- 2. Time-dependent hazard analyzing methods and models enable to capture changing hazard potentials. They further illustrate the different ways in which change can manifest itself, such as gradual versus abrupt changes, altered amplitudes, or seasonal shifts.
- Based on our experiences, we think that an over-simplification and/or a broad averaging over large scales in time and space will result in severe loss of information and hide the inherent mechanisms/reasons for changes. Consequently, this may yield to misleading system's analysis or even prognosis about the future development of such changes.
- 4. The symbiotic alliance between data science and earth science is a key determinant of successful timedependent natural hazard assessments.

Data Availability Statement

For the data used in Section 2, all 6-hourly surface wind speed data from the ERA-20C reanalysis (Poli et al., 2016) can be accessed from the European Centre for Medium-Range Weather Forecasts (ECMWF) portal (ECMWF, 2019). Daily precipitation sums for stations in Germany considered in Section 3 are provided via the open-data server of the German Weather Service (DWD, 2022). Discharge data presented in Section 4 were obtained from the Global Runoff Data Centre (GRDC, 2020). GLOFs, Section 5, were detected by Landsat, Shuttle Radar Topography Mission, and GTOPO30 data that are available from the US Geological Survey (earthexplorer.usgs.gov). Landsat images were preprocessed via the Earth Resources Observation and Science Center Science Processing Architecture ESPA (espa.cr.usgs.gov). Data and codes to estimate GLOF return periods are published by Veh, Korup, and Walz (2019) and available at GitHub (github.com/geveh/GLOFhazard). The data used in the Oklahoma case study in Section 6 can be obtained from Oklahoma Geological Survey (OGS, 2021) and the Oklahoma Corporation Commission (OCC, 2021).

Data in Sections 2, 3, 4 and 5 were analyzed and visualized using the open-source statistical programming software R (R Core Team, 2019). The simulations in Section 6 were conducted using R (R Core Team, 2019) and Python (Python Core Team, 2017). Figure 5 was generated using the Generic Mapping Tools (Wessel & Smith, 1998, 2007).

References

Addor, N., Rössler, O., Köplin, N., Huss, M., Weingartner, R., & Seibert, J. (2014). Robust changes and sources of uncertainty in the projected hydrological regimes of Swiss catchments. *Water Resources Research*, 50(10), 7541–7562. https://doi.org/10.1002/2014WR015549

- Alexander, L., Zhang, X., Peterson, T., Caesar, J., Gleason, B., Klein Tank, A., et al. (2006). Global observed changes in daily climate extremes of temperature and precipitation. *Journal of Geophysical Research*, 111(D5). https://doi.org/10.1029/2005JD006290
 - Allamano, P., Claps, P., & Laio, F. (2009a). An analytical model of the effects of catchment elevation on the flood frequency distribution. Water Resources Research, 45(1). https://doi.org/10.1029/2007WR006658
 - Allamano, P., Claps, P., & Laio, F. (2009b). Global warming increases flood risk in mountainous areas. *Geophysical Research Letters*, 36(24). https://doi.org/10.1029/2009GL041395
 - Amitrano, D. (2003). Brittle-ductile transition and associated seismicity: Experimental and numerical studies and relationship with the b value. Journal of Geophysical Research, 108(B1). https://doi.org/10.1029/2001JB000680
 - Arendt, A., Bliss, A., Bolch, T., Cogley, J., Gardner, A., & Hagen, J.-O. (2017). Randolph glacier inventory-a dataset of global glacier outlines: Version 6.0: Technical report, global land ice measurements from space. *RGI Consortium*. https://doi.org/10.7265/N5-RGI-60
 - Arheimer, B., Donnelly, C., & Lindström, G. (2017). Regulation of snow-fed rivers affects flow regimes more than climate change. Nature Communications, 8(1), 62. https://doi.org/10.1038/s41467-017-00092-8
 - Auer, I., Böhm, R., Jurković, A., Orlik, A., Potzmann, R., Schöner, W., et al. (2005). A new instrumental precipitation dataset for the greater alpine region for the period 1800–2002. International Journal of Climatology, 25(2), 139–166. https://doi.org/10.1002/joc.1135
 - Baker, T. (2017). New earthquake directive takes aim at future disposal rates. Oklahoma Corporation Commission. Retrieved from https://oklahoma.gov/content/dam/ok/en/occ/documents/ajls/news/2017/02-24-17-future-disposal.pdf
 - Barnett, T. P., Adam, J. C., & Lettenmaier, D. P. (2005). Potential impacts of a warming climate on water availability in snow-dominated regions. *Nature*, 438(7066), 303–309. https://doi.org/10.1038/nature04141
 - Befort, D. J., Wild, S., Kruschke, T., Ulbrich, U., & Leckebusch, G. C. (2016). Different long-term trends of extra-tropical cyclones and windstorms in era-20c and noaa-20cr reanalyses. Atmospheric Science Letters, 17(11), 586–595. https://doi.org/10.1002/asl.694
 - Begert, M., Schlegel, T., & Kirchhofer, W. (2005). Homogeneous temperature and precipitation series of Switzerland from 1864 to 2000. International Journal of Climatology, 25(1), 65–80. https://doi.org/10.1002/joc.1118

VOGEL ET AL.

The present work was developed within the framework of the Research Training Group "Natural Hazards and Risks in a Changing World" (NatRiskChange; GRK 2043/1) funded by the Deutsche Forschungsgemeinschaft (DFG). Open Access funding enabled and organized by Projekt DEAL.

- Belgiu, M., & Drăguţ, L. (2016). Random forest in remote sensing: A review of applications and future directions. *ISPRS Journal of Photo-grammetry and Remote Sensing*, 114, 24–31. https://doi.org/10.1016/j.isprsjprs.2016.01.011
- Beniston, M., Farinotti, D., Stoffel, M., Andreassen, L. M., Coppola, E., Eckert, N., et al. (2018). The European mountain cryosphere: A review of its current state, trends, and future challenges. *The Cryosphere*, 12(2), 759–794. https://doi.org/10.5194/tc-12-759-2018
- Beniston, M., Keller, F., & Goyette, S. (2003). Snow pack in the swiss alps under changing climatic conditions: An empirical approach for climate impacts studies. *Theoretical and Applied Climatology*, 74(1), 19–31. https://doi.org/10.1007/s00704-002-0709-1
- Berghuijs, W. R., Harrigan, S., Molnar, P., Slater, L. J., & Kirchner, J. W. (2019). The relative importance of different flood-generating mechanisms across Europe. *Water Resources Research*, 55(6), 4582–4593. https://doi.org/10.1029/2019WR024841
- Blöschl, G., Hall, J., Parajka, J., Perdigão, R. A. P., Merz, B., Arheimer, B., et al. (2017). Changing climate shifts timing of European floods. *Science*, 357(6351), 588–590. https://doi.org/10.1126/science.aan2506
- Blöschl, G., Hall, J., Viglione, A., Perdigão, R. A. P., Parajka, J., Merz, B., et al. (2019). Changing climate both increases and decreases european river floods. *Nature*, 573(7772), 108–111. https://doi.org/10.1038/s41586-019-1495-6
- Bolch, T., Shea, J. M., Liu, S., Azam, F. M., Gao, Y., & Gruber, S. (2019). Status and change of the cryosphere in the extended hindu kush himalaya region. In *The hindu kush Himalaya assessment* (pp. 209–255). Springer. https://doi.org/10.1007/978-3-319-92288-1_7
- Bosshard, T., Carambia, M., Goergen, K., Kotlarski, S., Krahe, P., Zappa, M., & Schär, C. (2013). Quantifying uncertainty sources in an ensemble of hydrological climate-impact projections. *Water Resources Research*, 49(3), 1523–1536. https://doi.org/10.1029/2011WR011533
- Breiman, L. (2001). Random forests. Machine Learning, 45(1), 5–32. https://doi.org/10.1023/A:1010933404324
 Bronstert, A., Agarwal, A., Boessenkool, B., Crisologo, I. M. F., Heistermann, M., et al. (2018). Forensic hydro-meteorological analysis of an extreme flash flood: The 2016-05-29 event in braunsbach, sw Germany. Science of the Total Environment, 630, 977–991. https://doi.org/10.1016/j.scitotenv.2018.02.241
- Bronstert, A., Bárdossy, A., Bismuth, C., Buiteveld, H., Disse, M., Engel, H., et al. (2007). Multi-scale modelling of land-use change and river training effects on floods in the Rhine basin. *River Research and Applications*, 23(10), 1102–1125. https://doi.org/10.1002/rra.1036
- Bronstert, A., Crisologo, I., Heistermann, M., Ozturk, U., Vogel, K., & Wendi, D. (2020). Flash-floods: More often, more severe, more damaging? An analysis of hydro-geo-environmental conditions and anthropogenic impacts. *Climate Change, Hazards and Adaptation Options*, 225–244. https://doi.org/10.1007/978-3-030-37425-9_12
- Bronstert, A., Niehoff, D., & Schiffler, G. R. (2023). Modelling infiltration and infiltration excess: The importance of fast and local processes. *Hydrological Processes*, 37(4), e14875. https://doi.org/10.1002/hyp.14875
- Brugnara, Y., Good, E., Squintu, A. A., van der Schrier, G., & Brönnimann, S. (2019). The EUSTACE global land station daily air temperature dataset. *Geoscience Data Journal*, 6(2), 189–204. https://doi.org/10.1002/gdj3.81
- Brun, F., Berthier, E., Wagnon, P., Kääb, A., & Treichler, D. (2017). A spatially resolved estimate of high mountain Asia glacier mass balances from 2000 to 2016. *Nature Geoscience*, 10(9), 668–673. https://doi.org/10.1038/ngco2999
- Brunner, M. I., & Naveau, P. (2023). Spatial variability in alpine reservoir regulation: Deriving reservoir operations from streamflow using generalized additive models. *Hydrology and Earth System Sciences*, 27(3), 673–687. https://doi.org/10.5194/hess-27-673-2023
- Bürger, G., Heistermann, M., & Bronstert, A. (2014). Towards sub-daily rainfall disaggregation via clausius clapeyron. Journal of Hydrometeorology, 15(3), 266–278. https://doi.org/10.1175/JHM-D-13-0161.1
- Bürger, G., Pfister, A., & Bronstert, A. (2019). Temperature-driven rise in extreme sub-hourly rainfall. Journal of Climate, 32(22), 7597–7609. https://doi.org/10.1175/JCLI-D-19-0136.1
- Bürger, G., Pfister, A., & Bronstert, A. (2021). Zunehmende Starkregenintensitäten als Folge der Klimaerwärmung. Datenanalyse und Zukunftsprojektion - Increasing intensity of heavy rainfall caused by global warming: Data analysis and future projections. *Hydrologie und Wasserbewirtschaftung*, 65(6), 262–271. https://doi.org/10.5675/HyWa_2021.6_1
- Byers, E., Gidden, M., Leclere, D., Balkovic, J., Burek, P., Ebi, K., et al. (2018). Global exposure and vulnerability to multi-sector development and climate change hotspots. *Environmental Research Letters*, 13(5). https://doi.org/10.1088/1748-9326/aabf45
- Carrivick, J. L., & Tweed, F. S. (2016). A global assessment of the societal impacts of glacier outburst floods. *Global and Planetary Change*, 144, 1–16. https://doi.org/10.1016/j.gloplacha.2016.07.001
- Chan, D., Kent, E. C., Berry, D. I., & Huybers, P. (2019). Correcting datasets leads to more homogeneous early-twentieth-century sea surface warming. *Nature*, *571*(7765), 393–397. https://doi.org/10.1038/s41586-019-1349-2
- Clague, J. J., & Evans, S. G. (2000). A review of catastrophic drainage of moraine-dammed lakes in british columbia. *Quaternary Science Reviews*, 19(17–18), 1763–1783. https://doi.org/10.1016/S0277-3791(00)00090-1
- Coles, S. G. (2001). An introduction to statistical modelling of extreme values. Springer. https://doi.org/10.1007/978-1-4471-3675-0
- Cook, S., & Quincey, D. (2015). Estimating the volume of alpine glacial lakes. Earth Surface Dynamics, 3(4), 559-575. https://doi.org/10.5194/
- esurf-3-559-2015 CRED & UNISDR. (2018). Economic losses, poverty & disasters: 1998-2017. Center for research on the Epidemiology of disasters & United Nations Office for disaster risk reduction.
- CRED & UNISDR. (2020). *Human cost of disasters. An overview of the last 20 years 2000-2019.* Center for Research on the Epidemiology of Disasters & United Nations Office for Disaster Risk Reduction.
- Derbile, E. K., & Kasei, R. A. (2012). Vulnerability of crop production to heavy precipitation in north-eastern Ghana. International Journal of Climate Change Strategies and Management, 4(1), 36–53. https://doi.org/10.1108/17568691211200209
- Dietze, M., anbd, U., Ozturk amd, K. L., Cook, R. B., Andermann, C., Beer, A., et al. (2022). More than heavy rain turning into fast-flowing water —A landscape perspective on the 2021 EIFEL floods. *Natural Hazards and Earth System Sciences*, 22(6), 1845–1856. https://doi.org/10.5194/ nhess-22-1845-2022
- Donat, M., Renggli, D., Wild, S., Alexander, L., Leckebusch, G., & Ulbrich, U. (2011). Reanalysis suggests long-term upward trends in European storminess since 1871. Geophysical Research Letters, 38(14). https://doi.org/10.1029/2011GL047995
- Donat, M. G., Lowry, A. L., Alexander, L. V., O'Gorman, P. A., & Maher, N. (2016). More extreme precipitation in the worlds dry and wet regions. *Nature Climate Change*, 6(5), 508–513. https://doi.org/10.1038/nclimate2941
- DWD. (2022). Deutscher Wetterdienst. [Dataset]. DWD. Retrieved from https://opendata.dwd.de/climate_environment/CDC/observations_germany/climate/daily/more_precip
- ECMWF. (2019). © 2019 European Centre for medium-range weather Forecasts [Dataset]. ECMWF. (Licence: CC BY 4.0). Retrieved from https://www.ecmwf.int/en/forecasts/dataset/ecmwf-reanalysis-20th-century-using-surface-observations-only
- EEA. (2017). Climate change, impacts and vulnerability in europe 2016 an indicator-based report (EEA report No. 1/2017). European Environment Agency. https://doi.org/10.2800/534806
- EEA. (2022). Annual economic damage caused by weather and climate-related extreme events in the eu member states (data visualization). European Environment Agency. Retrieved from https://www.eea.europa.eu/data-and-maps/daviz/natural-disasters-events-5/

Ellsworth, W. L. (2013). Injection-induced earthquakes. Science, 341(6142). https://doi.org/10.1126/science.1225942

Fauer, F. S., & Rust, H. W. (2023). Non-stationary large-scale statistics of precipitation extremes in central Europe. Stochastic Environmental Research and Risk Assessment, 1–13. https://doi.org/10.1007/s00477-023-02515-z

Fekete, B. M., Looser, U., Pietroniro, A., & Robarts, R. D. (2012). Rationale for monitoring discharge on the ground. Journal of Hydrometeorology, 13, 1977–1986. https://doi.org/10.1175/JHM-D-11-0126.1

- Fiedler, B., Hainzl, S., Zöller, G., & Holschneider, M. (2018). Detection of Gutenberg–Richter b-value changes in earthquake time series. Bulletin of the Seismological Society of America, 108(5A), 2778–2787. https://doi.org/10.1785/0120180091
- Fiedler, B., Zöller, G., Holschneider, M., & Hainzl, S. (2018). Multiple change-point detection in spatio-temporal seismicity data. Bulletin of the Seismological Society of America, 108(3A), 1147–1159. https://doi.org/10.1785/0120170236
- Fischer, M., Rust, H., & Ulbrich, U. (2019). A spatial and seasonal climatology of extreme precipitation return-levels: A case study. *Spatial Statistics*, *34*, 100275. https://doi.org/10.1016/j.spasta.2017.11.007
- Fischer, M., Rust, H. W., & Ulbrich, U. (2018). Seasonal cycle in German daily precipitation extremes. *Meteorologische Zeitschrift*, 27(1), 3–13. https://doi.org/10.1127/metz/2017/0845
- Franzke, C. (2012). Nonlinear trends, long-range dependence, and climate noise properties of surface temperature. *Journal of Climate*, 25(12), 4172–4183. https://doi.org/10.1175/JCLI-D-11-00293.1
- Frei, C., Davies, H. C., Gurtz, J., & Schär, C. (2000). Climate dynamics and extreme precipitation and flood events in Central Europe. Integrated Assessment, 1(4), 281–300. https://doi.org/10.1023/A:1018983226334
- Gall, M., Borden, K. A., & Cutter, S. L. (2009). When do losses count? Six fallacies of natural hazards loss data. Bulletin of the American Meteorological Society, 90(6), 799–810. https://doi.org/10.1175/2008BAMS2721.1
- Gariano, S. L., & Guzzetti, F. (2016). Landslides in a changing climate. *Earth-Science Reviews*, 162, 227–252. https://doi.org/10.1016/j.earscirev. 2016.08.011
- Gobiet, A., Kotlarski, S., Beniston, M., Heinrich, G., Rajczak, J., & Stoffel, M. (2014). 21st century climate change in the european alps A review. Science of the Total Environment, 493, 1138–1151. https://doi.org/10.1016/j.scitotenv.2013.07.050
- Goebel, T. H. W., Schorlemmer, D., Becker, T. W., Dresen, G., & Sammis, C. G. (2013). Acoustic emissions document stress changes over many seismic cycles in stick-slip experiments. *Geophysical Research Letters*, 40(10), 2049–2054. https://doi.org/10.1002/grl.50507
- Gorelick, N., Hancher, M., Dixon, M., Ilyushchenko, S., Thau, D., & Moore, R. (2017). Google earth engine: Planetary-scale geospatial analysis for everyone. *Remote Sensing of Environment*, 202, 18–27. https://doi.org/10.1016/j.rse.2017.06.031
- GRDC. (2020). Global runoff data Centre, 56068 Koblenz, Germany. [Dataset]. GRDC. Retrieved from https://portal.grdc.bafg.de
- Gudmundsson, L., Boulange, J., Do, H. X., Gosling, S. N., Grillakis, M. G., & Koutroulis, A. G. (2021). Globally observed trends in mean and extreme river flow attributed to climate change. *Science*, 371(6534), 1159–1162. https://doi.org/10.1126/science.aba3996
- Gueneralp, B., Gueneralp, I., & Liu, Y. (2015). Changing global patterns of urban exposure to flood and drought hazards. *Global Environmental Change*, 31, 217–225. https://doi.org/10.1016/j.gloenvcha.2015.01.002
- Gupta, A., & Baker, J. W. (2017). Estimating spatially varying event rates with a change point using bayesian statistics: Application to induced seismicity. *Structural Safety*, 65, 1–11. https://doi.org/10.1016/j.strusafe.2016.11.002
- Haeberli, W., Schaub, Y., & Huggel, C. (2017). Increasing risks related to landslides from degrading permafrost into new lakes in de-glaciating mountain ranges. *Geomorphology*, 293, 405–417. https://doi.org/10.1016/j.geomorph.2016.02.009
- Hainzl, S., Kraft, T., Wassermann, J., Igel, H., & Schmedes, E. (2006). Evidence for rainfall-triggered earthquake activity. *Geophysical Research Letters*, 33(19). https://doi.org/10.1029/2006GL027642
- Hallegatte, S., Green, C., Nicholls, R. J., & Corfee-Morlot, J. (2013). Future flood losses in major coastal cities. *Nature Climate Change*, 3(9), 802–806. https://doi.org/10.1038/nclimate1979
- Hammer, C., Fäh, D., & Ohrnberger, M. (2017). Automatic detection of wet-snow avalanche seismic signals. Natural Hazards: Journal of the International Society for the Prevention and Mitigation of Natural Hazards, 86(2), 601–618. https://doi.org/10.1007/s11069-016-2707-0
- Hanzer, F., Förster, K., Nemec, J., & Strasser, U. (2018). Projected cryospheric and hydrological impacts of 21st century climate change in the Ötztal Alps (Austria) simulated using a physically based approach. *Hydrology and Earth System Sciences*, 22(2), 1593–1614. https://doi.org/10. 5194/hess-22-1593-2018
- Harrison, S., Kargel, J. S., Huggel, C., Reynolds, J., Shugar, D. H., & Betts, R. A. (2018). Climate change and the global pattern of morainedammed glacial lake outburst floods. *The Cryosphere*, 12(4), 1195–1209. https://doi.org/10.5194/tc-12-1195-2018
- Hibert, C., Michéa, D., Provost, F., Malet, J.-P., & Geertsema, M. (2019). Exploration of continuous seismic recordings with a machine learning approach to document 20 yr of landslide activity in Alaska. *Geophysical Journal International*, 219(2), 1138–1147. https://doi.org/10.1093/gji/ ggz354
- Hoeppe, P. (2016). Trends in weather related disasters Consequences for insurers and society. Weather and Climate Extremes, 11, 70–79. https://doi.org/10.1016/j.wace.2015.10.002
- Huang, N. E., Shen, Z., Long, S. R., Wu, M. C., Shih, H. H., Zheng, Q., et al. (1998). The empirical mode decomposition and the hilbert spectrum for nonlinear and non-stationary time series analysis. *Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 454(1971), 903–995. https://doi.org/10.1098/rspa.1998.0193
- Hurst, H. E. (1951). Long-term storage capacity of reservoirs. Transactions of the American Society of Civil Engineers, 116(1), 770–799.
- Imoto, M. (1991). Changes in the magnitude-frequency b-value prior to large (m \geq 6.0) earthquakes in Japan. *Tectonophysics*, 193(4), 311–325. https://doi.org/10.1016/0040-1951(91)90340-X
- IPCC. (2012). Managing the risks of extreme events and disasters to advance climate change adaptation. In C. B. Field, V. Barros, T. F. Stocker, D. Qin, D. J. Dokken, K. L. Ebi, et al. (Eds.), A special report of working groups i and ii of the intergovernmental panel on climate change (p. 582). Cambridge University Press.
- IPCC. (2013). Climate change 2013: The physical science basis. In T. F. Stocker, D. Qin, G. K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), Contribution of working group i to the fifth assessment report of the intergovernmental panel on climate change (p. 1535). Cambridge University Press.
- IPCC. (2014). Climate change 2014: Impacts, adaptation, and vulnerability. Part A: Global and sectoral aspects. In C. B. Field, V. R. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, et al. (Eds.), *Contribution of working group ii to the fifth assessment report of the intergovernmental panel on climate change* (p. 1132). Cambridge University Press.
- IPCC. (2019a). In P. R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.-O. Pörtner, D. C. Roberts, et al. (Eds.), Climate change and land: An ipcc special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems. Intergovernmental Panel on Climate Change (IPCC).
- IPCC. (2019b). In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), IPCC special report on the ocean and cryosphere in a changing climate. Intergovernmental Panel on Climate Change (IPCC). https://doi.org/10.1017/9781009157964

- Jongman, B., Ward, P. J., & Aerts, J. C. (2012). Global exposure to river and coastal flooding: Long term trends and changes. *Global Environmental Change*, 22(4), 823–835. https://doi.org/10.1016/j.gloenvcha.2012.07.004
- Jongman, B., Winsemius, H. C., Aerts, J. C., Coughlan, E., Perez, D., & van Aalst, M. K. (2015). Declining vulnerability to river floods and the global benefits of adaptation. PNAS, 2271–2280. https://doi.org/10.1073/pnas.1414439112
- Kamer, Y., & Hiemer, S. (2013). Comment on "analysis of the b-values before and after the 23 october 2011 mw 7.2 van-ercis, Turkey, earthquake". *Tectonophysics*, 608, 1448–1451. https://doi.org/10.1016/j.tecto.2013.07.040
- Katz, R. W., Parlange, M. B., & Naveau, P. (2002). Statistics of extremes in hydrology. Advances in Water Resources, 25(8–12), 1287–1304. https://doi.org/10.1016/S0309-1708(02)00056-8
- Kemter, M., Merz, B., Marwan, N., Vorogushyn, S., & Blöschl, G. (2020). Joint trends in flood magnitudes and spatial extents across Europe. Geophysical Research Letters, 47(7), e2020GL087464. https://doi.org/10.1029/2020GL087464
- Knight, J., & Harrison, S. (2009). Sediments and future climate. Nature Geoscience, 2(4), 230. https://doi.org/10.1038/ngeo491
- Komori, J., Koike, T., Yamanokuchi, T., & Tshering, P. (2012). Glacial lake outburst events in the Bhutan himalayas. *Global Environmental Research*, 16(February 2012), 59–70.
- Konapala, G., Mishra, A. K., Wada, Y., & Mann, M. E. (2020). Climate change will affect global water availability through compounding changes in seasonal precipitation and evaporation. *Nature Communications*, 11(1), 1–10. https://doi.org/10.1038/s41467-020-16757-w
- Korup, O. (2021). Bayesian geomorphology. Earth Surface Processes and Landforms, 46(1), 151–172. https://doi.org/10.1002/esp.4995
- Korup, O., & Tweed, F. (2007). Ice, moraine, and landslide dams in mountainous terrain. *Quaternary Science Reviews*, 26(25–28), 3406–3422. https://doi.org/10.1016/j.quascirev.2007.10.012
 Kossin, J. P., Knapp, K. R., Olander, T. L., & Velden, C. S. (2020). Global increase in major tropical cyclone exceedance probability over the past
- four decades. Proceedings of the National Academy of Sciences, 117(22), 11975–11980. https://doi.org/10.1073/pnas.1920849117
- Kreibich, H., Thieken, A. H., Petrow, T., Müller, M., & Merz, B. (2005). Flood loss reduction of private households due to building precautionary measures – Lessons learned from the Elbe flood in August 2002. Natural Hazards and Earth System Sciences, 5(1), 117–126. https://doi.org/10. 5194/nhess-5-117-2005
- Kron, W., Bell, R., Thiebes, B., & Thieken, A. (2022). The July 2021 flood disaster in Germany. In (p. 12–44). High-level experts and leaders panel on water and disasters (HELP). Retrieved from https://www.wateranddisaster.org/
- Kruschke, J. K. (2015). Doing bayesian data analysis: A tutorial with R, JAGS, and Stan. Academic Press. Retrieved from https://nyu-cdsc.github.io/learningr/asset/kruschke_bayesian_in_R.pdf

Kulhanek, O. (2005). Seminar on b-value. Dept. of Geophysics (pp. 10-190). Charles University.

- Kundzewicz, Z. W., Kanae, S., Seneviratne, S. I., Handmer, J., Nicholls, N., Peduzzi, P., et al. (2014). Flood risk and climate change: Global and regional perspectives. *Hydrological Sciences Journal*, 59(1), 1–28. https://doi.org/10.1080/02626667.2013.857411
- Langenbruch, C., Weingarten, M., & Zoback, M. D. (2018). Physics-based forecasting of man-made earthquake hazards in Oklahoma and Kansas. *Nature Communications*, 9(1), 1–10. https://doi.org/10.1038/s41467-018-06167-4
- Langenbruch, C., & Zoback, M. D. (2016). How will induced seismicity in Oklahoma respond to decreased saltwater injection rates? *Science Advances*, 2(11), e1601542. https://doi.org/10.1126/sciadv.1601542
- Laternser, M., & Schneebeli, M. (2003). Long-term snow climate trends of the Swiss Alps (1931–99). International Journal of Climatology, 23(7), 733–750. https://doi.org/10.1002/joc.912
- Leckebusch, G. C., Weimer, A., Pinto, J. G., Reyers, M., & Speth, P. (2008). Extreme wind storms over europe in present and future climate: A cluster analysis approach. *Meteorologische Zeitschrift*, 17(1), 67–82. https://doi.org/10.1127/0941-2948/2008/0266
- Lehmann, J., Coumou, D., & Frieler, K. (2015). Increased record-breaking precipitation events under global warming. *Climatic Change*, 132(4), 501–515. https://doi.org/10.1007/s10584-015-1434-y
- Leinss, S., Bernardini, E., Jacquemart, M., & Dokukin, M. (2020). Glacier detachments and rock-ice avalanches in the Petra Pervogo range, Tajikistan (1973–2019). Natural Hazards and Earth System Sciences, 1–31. https://doi.org/10.5194/nhess-2020-285
- Luukko, P. J. J., Helske, J., & Räsänen, E. (2016). Introducing libeemd: A program package for performing the ensemble empirical mode decomposition. *Computational Statistics*, 31(2), 545–557. https://doi.org/10.1007/s00180-015-0603-9
- Maharjan, S. B., Mool, P., Lizong, W., Xiao, G., Shrestha, F., & Shrestha, R. (2018). The status of glacial lakes in the Hindu Kush himalaya -ICIMOD research report 2018/1. International Centre for Integrated Mountain Development (ICIMOD). https://doi.org/10.53055/ ICIMOD.742
- Maraun, D., Rust, H. W., & Osborn, T. J. (2009). The annual cycle of heavy precipitation across the UK: A model based on extreme value statistics. *Journal of Climatology*, 29(12), 1731–1744. https://doi.org/10.1002/joc.1811
- Marone, C. (2018). Training machines in earthly ways. Nature Geoscience, 11(5), 301-302. https://doi.org/10.1038/s41561-018-0117-5
- Maton, C. (2018). Regime shift of snow days in Switzerland. Geophysical Research Letters, 35(12). https://doi.org/10.1029/2008GL033998
- Marty, C., Schlögl, S., Bavay, M., & Lehning, M. (2017). How much can we save? Impact of different emission scenarios on future snow cover in the Alps. *The Cryosphere*, 11(1), 517–529. https://doi.org/10.5194/tc-11-517-2017
- Marwan, N. (2019). Recurrence plot techniques for the investigation of recurring phenomena in the system earth (Doctoral dissertation). https:// doi.org/10.25932/publishup-44197
- Maurer, J. M., Schaefer, J., Rupper, S., & Corley, A. (2019). Acceleration of ice loss across the himalayas over the past 40 years. *Science Advances*, 5(6), eaav7266. https://doi.org/10.1126/sciadv.aav7266
- Meile, T., Boillat, J.-L., & Schleiss, A. J. (2011). Hydropeaking indicators for characterization of the upper-Rhone river in Switzerland. Aquatic Sciences, 73(1), 171–182. https://doi.org/10.1007/s00027-010-0154-7
- Merz, B., Hall, J., Disse, M., & Schumann, A. (2010). Fluvial flood risk management in a changing world. Natural Hazards and Earth System Sciences, 10(3), 509–527. https://doi.org/10.5194/nhess-10-509-2010
- Merz, R., & Blöschl, G. (2003). A process typology of regional floods. Water Resources Research, 39(12). https://doi.org/10.1029/ 2002WR001952
- Mínguez, R., Méndez, F., Izaguirre, C., Menéndez, M., & Losada, I. J. (2010). Pseudo-optimal parameter selection of non-stationary generalized extreme value models for environmental variables. *Environmental Modelling & Software*, 25(12), 1592–1607. https://doi.org/10.1016/j. envsoft.2010.05.008
- Montoya-Noguera, S., & Wang, Y. (2017). Bayesian identification of multiple seismic change points and varying seismic rates caused by induced seismicity. *Geophysical Research Letters*, 44(8), 3509–3516. https://doi.org/10.1002/2016GL072266
- Moran, T. (2021). Factors affecting the severity of european winter windstorms (Doctoral dissertation, FU Berlin). https://doi.org/10.17169/ refubium-29369
- Mueller, E. N., & Pfister, A. (2011). Increasing occurrence of high-intensity rainstorm events relevant for the generation of soil erosion in a temperate lowland region in Central Europe. Journal of Hydrology, 411(3), 266–278. https://doi.org/10.1016/j.jhydrol.2011.10.005



- MunichRE. (2015). Loss events in Europe 1980–2014: 10 costliest winter storms ordered by insured losses (tech. Rep.). Tech. Rep., Münchener Rückversicherungs-Gesellschaft, Geo risks research, NatCatSERVICE.
- Musselman, K. N., Clark, M. P., Liu, C., Ikeda, K., & Rasmussen, R. (2017). Slower snowmelt in a warmer world. Nature Climate Change, 7(3), 214–219. https://doi.org/10.1038/nclimate3225
- Nakaya, S. (2006). Spatiotemporal variation in b value within the subducting slab prior to the 2003 tokachi-oki earthquake (m 8.0), Japan. Journal of Geophysical Research, 111(B3). https://doi.org/10.1029/2005JB003658
- Nanjo, K., Hirata, N., Obara, K., & Kasahara, K. (2012). Decade-scale decrease in b value prior to the M9-class 2011 Tohoku and 2004 Sumatra quakes. *Geophysical Research Letters*, 39(20). https://doi.org/10.1029/2012GL052997
- Nie, Y., Liu, Q., Wang, J., Zhang, Y., Sheng, Y., & Liu, S. (2018). An inventory of historical glacial lake outburst floods in the Himalayas based on remote sensing observations and geomorphological analysis. *Geomorphology*, *308*, 91–106. https://doi.org/10.1016/j.geomorph.2018.02.002
- Nie, Y., Sheng, Y., Liu, Q., Liu, L., Liu, S., Zhang, Y., & Song, C. (2017). A regional-scale assessment of Himalayan glacial lake changes using satellite observations from 1990 to 2015. *Remote Sensing of Environment*, 189, 1–13. https://doi.org/10.1016/j.rse.2016.11.008
- Niehoff, D., Fritsch, U., & Bronstert, A. (2002). Land-use impacts on storm-runoff generation: Scenarios of land-use change and simulation of hydrological response in a meso-scale catchment in sw-Germany. *Journal of Hydrology*, 267(1–2), 80–93. https://doi.org/10.1016/S0022-1694 (02)00142-7
- Nuannin, P., Kulhanek, O., & Persson, L. (2005). Spatial and temporal b value anomalies preceding the devastating off coast of NW sumatra earthquake of december 26, 2004. *Geophysical Research Letters*, 32(11). https://doi.org/10.1029/2005GL022679
- OCC. (2021). Oklahoma corporation commission. [Dataset]. OCC. Retrieved from https://oklahoma.gov/occ/divisions/oil-gas/oil-gas-data.html OGS. (2021). Oklahoma geological survey. [Dataset]. OGS. Retrieved from http://www.ou.edu/ogs/research/earthquakes/catalogs.html
- Osti, R., & Egashira, S. (2009). Hydrodynamic characteristics of the tam pokhari glacial lake outburst flood in the mt. everest region, Nepal. Hydrological Processes: An International Journal, 23(20), 2943–2955. https://doi.org/10.1002/hyp.7405
- Otto, F. E., Harrington, L. J., Frame, D., Boyd, E., Lauta, K. C., Wehner, M., et al. (2020). Toward an inventory of the impacts of human-induced climate change. *Bulletin of the American Meteorological Society*, 101(11), E1972–E1979. https://doi.org/10.1175/BAMS-D-20-0027.1
- Otto, I. M., Reckien, D., Reyer, C. P., Marcus, R., Le Masson, V., Jones, L., et al. (2017). Social vulnerability to climate change: A review of concepts and evidence. *Regional Environmental Change*, 17(6), 1651–1662. https://doi.org/10.1007/s10113-017-1105-9
- Öztürk, U. (2018), Learning more to predict landslides (doctoralthesis, Universität Potsdam), https://doi.org/10.25932/publishup-42643
- Palutikof, J., & Skellern, A. (1991). Storm severity over britain, a report to commercial union general insurance, climatic research unit. University of East Anglia.
- Peduzzi, P., Dao, Q.-H., Herold, C., Rochette, D. S., & Sanahuja, H. (2001). Feasibility study report on global risk and vulnerability index-trends per year (gravity). United Nations Development. Retrieved from http://archive-ouverte.unige.ch/unige:32337
- Pekel, J.-F., Cottam, A., Gorelick, N., & Belward, A. S. (2016). High-resolution mapping of global surface water and its long-term changes. *Nature*, 540(7633), 418–422. https://doi.org/10.1038/nature20584
- Pérez Ciria, T., Labat, D., & Chiogna, G. (2019). Detection and interpretation of recent and historical streamflow alterations caused by river damming and hydropower production in the Adige and Inn river basins using continuous, discrete and multiresolution wavelet analysis. *Journal* of Hydrology, 578, 124021. https://doi.org/10.1016/j.jhydrol.2019.124021
- Peter, M., Rust, H. W., & Ulbrich, U. (2024). Interannual variations in the seasonal cycle of extreme precipitation in Germany and the response to climate change. *Natural Hazards and Earth System Sciences*, 24(4), 1261–1285. https://doi.org/10.5194/nhess-24-1261-2024
- Petrow, T., & Merz, B. (2009). Trends in flood magnitude, frequency and seasonality in Germany in the period 1951–2002. *Journal of Hydrology*, 371(1), 129–141. https://doi.org/10.1016/j.jhydrol.2009.03.024
- Pfister, L., Kwadijk, J., Musy, A., Bronstert, A., & Hoffmann, L. (2004). Climate change, land use change and runoff prediction in the Rhine-Meuse basins. *River Research and Applications*, 20(3), 229–241. https://doi.org/10.1002/tra.775
- Pirret, J. S., Knippertz, P., & Trzeciak, T. M. (2017). Drivers for the deepening of severe european windstorms and their impacts on forecast quality. *Quarterly Journal of the Royal Meteorological Society*, 143(702), 309–320. https://doi.org/10.1002/qj.2923
- Plate, E. J. (2002). Flood risk and flood management. Journal of Hydrology, 267(1-2), 2-11. https://doi.org/10.1016/S0022-1694(02)00135-X
- Poli, P., Hersbach, H., Dee, D. P., Berrisford, P., Simmons, A. J., & Vitart, F. (2016). ERA-20C: An atmospheric reanalysis of the twentieth century. Journal of Climate, 29(11), 4083–4097. https://doi.org/10.1175/JCLI-D-15-0556.1
- Python Core Team. (2017). Python: A dynamic, open source programming language (Python version 3.4.6). [Computer software manual]. Python Core Team. Retrieved from https://www.python.org/
- R Core Team. (2019). R: A language and environment for statistical computing. [Computer software manual]. R Core Team. Retrieved from https://www.R-project.org/
- Reichstein, M., Camps-Valls, G., Stevens, B., Jung, M., Denzler, J., & Carvalhais, N., & Prabhat. (2019). Deep learning and process understanding for data-driven Earth system science. *Nature*, 566(7743), 195–204. https://doi.org/10.1038/s41586-019-0912-1
- Richardson, S. D., & Reynolds, J. M. (2000). An overview of glacial hazards in the Himalayas. *Quaternary International*, 65, 31–47. https://doi.org/10.1016/S1040-6182(99)00035-X
- Rohat, G., Flacke, J., Dosio, A., Dao, H., & van Maarseveen, M. (2019). Projections of human exposure to dangerous heat in African cities under multiple socioeconomic and climate scenarios. *Earth's Future*, 7(5), 528–546. https://doi.org/10.1029/2018EF001020
- Rohat, G., Flacke, J., Dosio, A., Pedde, S., Dao, H., & van Maarseveen, M. (2019). Influence of changes in socioeconomic and climatic conditions on future heat-related health challenges in europe. *Global and Planetary Change*, 172, 45–59. https://doi.org/10.1016/j.gloplacha.2018.09.013
- Rosenzweig, C., Tubiello, F. N., Goldberg, R., Mills, E., & Bloomfield, J. (2002). Increased crop damage in the US from excess precipitation under climate change. *Global Environmental Change*, *12*(3), 197–202. https://doi.org/10.1016/S0959-3780(02)0008-0
- Rössler, O., Froidevaux, P., Börst, U., Rickli, R., Martius, O., & Weingartner, R. (2014). Retrospective analysis of a nonforecasted rain-on-snow flood in the alps – A matter of model limitations or unpredictable nature? *Hydrology and Earth System Sciences*, 18(6), 2265–2285. https://doi. org/10.5194/hess-18-2265-2014
- Rottler, E., Bronstert, A., Bürger, G., & Rakovec, O. (2020). Projected changes in Rhine river flood seasonality under global warming. Hydrology and Earth System Sciences Discussions, 2020, 1–25. https://doi.org/10.5194/hess-2020-605
- Rottler, E., Francke, T., Bürger, G., & Bronstert, A. (2020). Long-term changes in central European river discharge for 1869–2016: Impact of changing snow covers, reservoir constructions and an intensified hydrological cycle. *Hydrology and Earth System Sciences*, 24(4), 1721–1740. https://doi.org/10.5194/hess-24-1721-2020
- Rottler, E., Kormann, C., Francke, T., & Bronstert, A. (2019). Elevation-dependent warming in the Swiss Alps 1981–2017: Features, forcings and feedbacks. *International Journal of Climatology*, 39(5), 2556–2568. https://doi.org/10.1002/joc.5970
- Rust, H. (2007). Detection of long-range dependence: Applications in climatology and hydrology (doctoral thesis). Universität Potsdam.

- Rust, H., Maraun, D., & Osborn, T. J. (2009). Modelling seasonality in extreme rainfall: A UK case study. European Physical Journal: Special Topics, 174, 99–111. https://doi.org/10.1140/epjst/e2009-01093-7
- Rypdal, K., Østvand, L., & Rypdal, M. (2013). Long-range memory in earth's surface temperature on time scales from months to centuries. Journal of Geophysical Research: Atmospheres, 118(13), 7046–7062. https://doi.org/10.1002/jgrd.50399
- Salerno, F., Guyennon, N., Thakuri, S., Viviano, G., Romano, E., & Vuillermoz, E. (2015). Weak precipitation, warm winters and springs impact glaciers of south slopes of Mt. Everest (central Himalaya) in the last 2 decades (1994–2013). The Cryosphere, 9(3), 1229–1247. https://doi.org/ 10.5194/tc-9-1229-2015
- Scherrer, S. C., Appenzeller, C., & Laternser, M. (2004). Trends in Swiss Alpine snow days: The role of local- and large-scale climate variability. *Geophysical Research Letters*, 31(13). https://doi.org/10.1029/2004GL020255
- Scherrer, S. C., Begert, M., Croci-Maspoli, M., & Appenzeller, C. (2016). Long series of Swiss seasonal precipitation: Regionalization, trends and influence of large-scale flow. *International Journal of Climatology*, 36(11), 3673–3689. https://doi.org/10.1002/joc.4584
- Scherrer, S. C., Wüthrich, C., Croci-Maspoli, M., Weingartner, R., & Appenzeller, C. (2013). Snow variability in the swiss alps 1864–2009. International Journal of Climatology, 33(15), 3162–3173. https://doi.org/10.1002/joc.3653
- Schmidli, J., & Frei, C. (2005). Trends of heavy precipitation and wet and dry spells in Switzerland during the 20th century. International Journal of Climatology, 25(6), 753–771. https://doi.org/10.1002/joc.1179
- Scholz, C. (1968). The frequency-magnitude relation of microfracturing in rock and its relation to earthquakes. Bulletin of the Seismological Society of America, 58(1), 399–415. https://doi.org/10.1785/BSSA0580010399
- Scholz, C. H. (2015). On the stress dependence of the earthquake b value. Geophysical Research Letters, 42(5), 1399–1402. https://doi.org/10. 1002/2014GL062863
- Scholz, C. H., Tan, Y. J., & Albino, F. (2019). The mechanism of tidal triggering of earthquakes at mid-ocean ridges. Nature Communications, 10(2526). https://doi.org/10.1038/s41467-019-10605-2
- Schorlemmer, D., Wiemer, S., & Wyss, M. (2005). Variations in earthquake-size distribution across different stress regimes. *Nature*, 437(7058), 539–542. https://doi.org/10.1038/nature04094
- Smith, W. D. (1981). The b-value as an earthquake precursor. Nature, 289(5794), 136-139. https://doi.org/10.1038/289136a0
- Spada, M., Tormann, T., Wiemer, S., & Enescu, B. (2013). Generic dependence of the frequency-size distribution of earthquakes on depth and its relation to the strength profile of the crust. *Geophysical Research Letters*, 40(4), 709–714. https://doi.org/10.1029/2012GL054198
- Stewart, I. T. (2009). Changes in snowpack and snowmelt runoff for key mountain regions. *Hydrological Processes*, 23(1), 78–94. https://doi.org/ 10.1002/hyp.7128
- Stott, P. A., Christidis, N., Otto, F. E., Sun, Y., Vanderlinden, J. P., van Oldenborgh, G. J., et al. (2016). Attribution of extreme weather and climate-related events. Wiley Interdisciplinary Reviews: Climate Change, 7(1), 23–41. https://doi.org/10.1002/wcc.380
- Thieken, A. H., Bubeck, P., Heidenreich, A., von Keyserlingk, J., Dillenardt, L., & Otto, A. (2022). Performance of the flood warning system in Germany in July 2021 Insights from affected residents. *EGUsphere*, 1–26. https://doi.org/10.5194/egusphere-2022-244
- Thieken, A. H., Kienzler, S., Kreibich, H., Kuhlicke, C., Kunz, M., Mühr, B., et al. (2016). Review of the flood risk management system in Germany after the major flood in 2013. *Ecology and Society*, 21(2). https://doi.org/10.5751/ES-08547-210251
- Ulbrich, U., Brücher, T., Fink, A. H., Leckebusch, G. C., Krüger, A., & Pinto, J. G. (2003). The central European floods of August 2002: Part 1 Rainfall periods and flood development. Weather, 58(10), 371–377. https://doi.org/10.1256/wea.61.03A
- Ulbrich, U., Leckebusch, G., & Pinto, J. G. (2009). Extra-tropical cyclones in the present and future climate: A review. *Theoretical and Applied Climatology*, 96(1), 117–131. https://doi.org/10.1007/s00704-008-0083-8
- Ulrich, J., Jurado, O. E., Peter, M., Scheibel, M., & Rust, H. W. (2020). Estimating idf curves consistently over durations with spatial covariates. *Water*, 12(11). https://doi.org/10.3390/w12113119
- UNDRR. (2019). Global assessment report on disaster risk reduction 2019 (Assessment report). United Nations Office for Disaster Risk Reduction. Retrieved from https://www.undrr.org/publication/global-assessment-report-disaster-risk-reduction-2019
- UNISDR. (2011). Global Assessment report on disaster risk reduction. Revealing Risk, redefining development. In (p. 156). United Nations Office for Disaster Risk Reduction. Retrieved from https://www.undrr.org/publication/global-assessment-report-disaster-risk-reduction-2011 UNISDR. (2015). Sendai framework for disaster risk reduction 2015-2030. In 3rd united nations world conference on disaster risk reduction.
- CNISDR. (2015). Sendal framework for disaster risk reduction 2015-2050. In *Sra united nations world conference on alsoster risk reduction*. Retrieved from https://www.undrr.org/publication/sendai-framework-disaster-risk-reduction-2015-2030
- Urbancic, T., Trifu, C.-I., Long, J., & Young, R. P. (1992). Space-time correlations of b values with stress release. Pure and Applied Geophysics, 139(3), 449–462. https://doi.org/10.1007/BF00879946
- van Stiphout, T., Zhuang, J., & Marsan, D. (2012). Seismicity declustering. Community Online Resource for Statistical Seismicity Analysis, 10(1). https://doi.org/10.5078/corssa-52382934
- Veh, G., Korup, O., Roessner, S., & Walz, A. (2018). Detecting himalayan glacial lake outburst floods from landsat time series. *Remote Sensing of Environment*, 207, 84–97. https://doi.org/10.1016/j.rse.2017.12.025
- Veh, G., Korup, O., von Specht, S., Roessner, S., & Walz, A. (2019). Unchanged frequency of moraine-dammed glacial lake outburst floods in the himalaya. Nature Climate Change, 9(5), 379–383. https://doi.org/10.1038/s41558-019-0437-5
- Veh, G., Korup, O., & Walz, A. (2019). Supplementary data: Hazard from Himalyan glacier lake outburst floods. [Dataset]. Zenodo. https://doi. org/10.5281/zenodo.3523213
- Veh, G., Korup, O., & Walz, A. (2020). Hazard from himalayan glacier lake outburst floods. Proceedings of the National Academy of Sciences, 117(2), 907–912. https://doi.org/10.1073/pnas.1914898117
- Verbunt, M., Zwaaftink, M. G., & Gurtz, J. (2005). The hydrologic impact of land cover changes and hydropower stations in the Alpine Rhine basin. *Ecological Modelling*, 187(1), 71–84. https://doi.org/10.1016/j.ecolmodel.2005.01.027
- Vergara-Temprado, J., Ban, N., & Schär, C. (2021). Extreme sub-hourly precipitation intensities scale close to the clausius-clapeyron rate over europe. *Geophysical Research Letters*, 48(3). https://doi.org/10.1029/2020GL089506
- Viglione, A., Merz, R., Salinas, J. L., & Blöschl, G. (2013). Flood frequency hydrology: 3. A bayesian analysis. Water Resources Research, 49(2), 675–692. https://doi.org/10.1029/2011WR010782
- Vincent, L. A., Zhang, X., Bonsal, B. R., & Hogg, W. D. (2002). Homogenization of daily temperatures over Canada. *Journal of Climate*, 15(11), 1322–1334. https://doi.org/10.1175/1520-0442(2002)015(1322:HODTOC)2.0.CO;2
- Viviroli, D., Archer, D. R., Buytaert, W., Fowler, H. J., Greenwood, G. B., Hamlet, A. F., et al. (2011). Climate change and mountain water resources: Overview and recommendations for research, management and policy. *Hydrology and Earth System Sciences*, 15(2), 471–504. https://doi.org/10.5194/hess-15-471-2011
- Vogel, J., Paton, E., Aich, V., & Bronstert, A. (2021). Increasing compound warm spells and droughts in the mediterranean basin. *Weather and Climate Extremes*, *32*, 100312. https://doi.org/10.1016/j.wace.2021.100312



- Vogel, K., Weise, L., Schröter, K., & Thieken, A. H. (2018). Identifying driving factors in flood-damaging processes using graphical models. Water Resources Research, 54(11), 8864–8889. https://doi.org/10.1029/2018WR022858
- Vormoor, K., Lawrence, D., Heistermann, M., & Bronstert, A. (2015). Climate change impacts on the seasonality and generation processes of floods–projections and uncertainties for catchments with mixed snowmelt/rainfall regimes. *Hydrology and Earth System Sciences*, 19(2), 913–931. https://doi.org/10.5194/hess-19-913-2015
- Vormoor, K., Lawrence, D., Schlichting, L., Wilson, D., & Wong, W. K. (2016). Evidence for changes in the magnitude and frequency of observed rainfall vs. snowmelt driven floods in Norway. *Journal of Hydrology*, 538, 33–48. https://doi.org/10.1016/j.jhydrol.2016.03.066
- Vorogushyn, S., Apel, H., Kemter, M., & Thieken, A. H. (2022). Analyse der Hochwassergefährdung im Ahrtal unter Berücksichtigung historischer Hochwasser. *Hydrologie und Wasserbewirtschaftung*, 66(5), 244–254. https://doi.org/10.5675/HyWa_2021.5_2
- Walder, J. S., & O'Connor, J. E. (1997). Methods for predicting peak discharge of floods caused by failure of natural and constructed earthen dams. Water Resources Research, 33(10), 2337–2348. https://doi.org/10.1029/97WR01616
- Wang, X., Guo, X., Yang, C., Liu, Q., Wei, J., Zhang, Y., et al. (2020). Glacial lake inventory of high-mountain asia in 1990 and 2018 derived from landsat images. *Earth System Science Data*, 12(3), 2169–2182. https://doi.org/10.5194/essd-12-2169-2020
- Watanbe, T., & Rothacher, D. (1996). The 1994 lugge tsho glacial lake outburst flood, Bhutan himalaya. *Mountain Research and Development*, 16(1), 77–81. https://doi.org/10.2307/3673897
- Wendi, D., Merz, B., & Marwan, N. (2019). Assessing hydrograph similarity and rare runoff dynamics by cross recurrence plots. Water Resources Research, 55(6), 4704–4726. https://doi.org/10.1029/2018WR024111
- Wessel, P., & Smith, W. H. F. (1998). New, improved version of generic mapping tools released. *Eos, Transactions American Geophysical Union*, 79(47), 579. https://doi.org/10.1029/98EO00426
- Wessel, P., & Smith, W. H. F. (2007). The Generic Mapping Tools (GMT) (version 4.2.1). [Computer software manual]. www.soest.hawaii. edu/gmt
- Wetter, O., Pfister, C., Weingartner, R., Luterbacher, J., Reist, T., & Trösch, J. (2011). The largest floods in the High Rhine basin since 1268 assessed from documentary and instrumental evidence. *Hydrological Sciences Journal*, 56(5), 733–758. https://doi.org/10.1080/02626667. 2011.583613
- Wildenhahn, E., & Klaholz, U. (1996). Große Speicherseen im Einzugsgebiet des Rheins (Tech. Rep. No. Bericht Nr. II-10). Koblenz: Internationale Kommission fur die Hydrologie des Rheingebietes (KHR).
- Willems, P. (2013). Multidecadal oscillatory behaviour of rainfall extremes in europe. Climatic Change, 120(4), 931–944. https://doi.org/10.1007/s10584-013-0837-x
- Wirtz, A., Kron, W., Löw, P., & Steuer, M. (2014). The need for data: Natural disasters and the challenges of database management. Natural Hazards, 70(1), 135–157. https://doi.org/10.1007/s11069-012-0312-4
- Wu, Z., Huang, N. E., Long, S. R., & Peng, C.-K. (2007). On the trend, detrending, and variability of nonlinear and nonstationary time series. Proceedings of the National Academy of Sciences, 104(38), 14889–14894. https://doi.org/10.1073/pnas.0701020104
- Wyss, M. (1973). Towards a physical understanding of the earthquake frequency distribution. *Geophysical Journal International*, 31(4), 341–359. https://doi.org/10.1111/j.1365-246X.1973.tb06506.x
- Xu, D. (1988). Characteristics of debris flow caused by outburst of glacial lake in boqu river, xizang, China, 1981. *Geojournal*, 17(4), 569–580. https://doi.org/10.1007/BF00209443
- Zeder, J., & Fischer, E. M. (2020). Observed extreme precipitation trends and scaling in central europe. Weather and Climate Extremes, 29, 100266. https://doi.org/10.1016/j.wace.2020.100266
- Zolina, O., Simmer, C., Kapala, A., Bachner, S., Gulev, S., & Maechel, H. (2008). Seasonally dependent changes of precipitation extremes over Germany since 1950 from a very dense observational network. *Journal of Geophysical Research*, 113(D6). https://doi.org/10.1029/ 2007JD008393
- Zscheischler, J., Westra, S., Van Den Hurk, B. J., Seneviratne, S. I., Ward, P. J., Pitman, A., et al. (2018). Future climate risk from compound events. *Nature Climate Change*, 8(6), 469–477. https://doi.org/10.1038/s41558-018-0156-3
- Zubler, E. M., Scherrer, S. C., Croci-Maspoli, M., Liniger, M. A., & Appenzeller, C. (2014). Key climate indices in Switzerland; expected changes in a future climate. *Climatic Change*, 123(2), 255–271. https://doi.org/10.1007/s10584-013-1041-8